

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bustelo, Monserrat; Flabbi, Luca; Viollaz, Mariana

Working Paper The gender labor market gap in the digital economy

IDB Working Paper Series, No. IDB-WP-1056

Provided in Cooperation with: Inter-American Development Bank (IDB), Washington, DC

Suggested Citation: Bustelo, Monserrat; Flabbi, Luca; Viollaz, Mariana (2019) : The gender labor market gap in the digital economy, IDB Working Paper Series, No. IDB-WP-1056, Inter-American Development Bank (IDB), Washington, DC, https://doi.org/10.18235/0001941

This Version is available at: https://hdl.handle.net/10419/208201

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/3.0/igo/legalcode







IDB WORKING PAPER SERIES Nº IDB-WP-01056

The Gender Labor Market Gap in the Digital Economy

Monserrat Bustelo Luca Flabbi Mariana Viollaz

Inter-American Development Bank Social Sector

October 2019



The Gender Labor Market Gap in the Digital Economy

Monserrat Bustelo Luca Flabbi Mariana Viollaz Cataloging-in-Publication data provided by the Inter-American Development Bank Felipe Herrera Library Bustelo, Monserrat.

The gender labor market gap in the digital economy / Monserrat Bustelo, Luca Flabbi, Mariana Viollaz.

p. cm. — (IDB Working Paper ; 1056)

Includes bibliographic references.

1. Women-Employment-Effect of technological innovations on-Latin America. 2. Sex discrimination in employment-Latin America. 3. Information technology-Economic aspects-Latin America. I. Flabbi, Luca. II. Viollaz, Mariana. III. Inter-American Development Bank. Gender and Diversity Division. IV. Title. V. Series. IDB-WP-1056

JEL Codes: J16, J31 Keywords: Digital economy, STEM, Gender wage gap, Return to skills

http://www.iadb.org

Copyright © 2019 Inter-American Development Bank. This work is licensed under a Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives (CC-IGO BY-NC-ND 3.0 IGO) license (<u>http://creativecommons.org/licenses/by-nc-nd/3.0/igo/</u> <u>legalcode</u>) and may be reproduced with attribution to the IDB and for any non-commercial purpose, as provided below. No derivative work is allowed.

Any dispute related to the use of the works of the IDB that cannot be settled amicably shall be submitted to arbitration pursuant to the UNCITRAL rules. The use of the IDB's name for any purpose other than for attribution, and the use of IDB's logo shall be subject to a separate written license agreement between the IDB and the user and is not authorized as part of this CC-IGO license.

Following a peer review process, and with previous written consent by the Inter-American Development Bank (IDB), a revised version of this work may also be reproduced in any academic journal, including those indexed by the American Economic Association's EconLit, provided that the IDB is credited and that the author(s) receive no income from the publication. Therefore, the restriction to receive income from such publication shall only extend to the publication's author(s). With regard to such restriction, in case of any inconsistency between the Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives license and these statements, the latter shall prevail.

Note that link provided above includes additional terms and conditions of the license.

The opinions expressed in this publication are those of the authors and do not necessarily reflect the views of the Inter-American Development Bank, its Board of Directors, or the countries they represent.



The Gender Labor Market Gap in the Digital Economy*

Monserrat Bustelo

Luca Flabbi

Mariana Viollaz

October, 2019

Abstract

Recent years have seen an ever-greater expansion of the *digital economy*, a development that may bring new opportunities to workers who were at a disadvantage in the traditional economy. We focus on a specific set of workers who belong to such a group: women. We study a skill set of particular relevance in the digital economy and estimate their returns in the labor market, according to gender, across four Latin American countries. We find that information and communication technologies (ICT) skills and science, technology, engineering, and mathematics (STEM) skills yield significant positive returns for both men and women. However, there is a significant gender gap that favors men on the STEM returns. There is also a sizable gender gap regarding the amount of skills accumulated by gender. Through an Oaxaca-Blinder decomposition, we estimate that up to 80% of the gender gap in hourly wages may be due to the lower returns that women receive, relative to men, on their STEM skills. If an investment in skills relevant to the digital economy may be beneficial for the labor market performance of both men and women, why returns to STEM exhibit such strong gender asymmetries remains an open and relevant question.

^{*} Contact Information:

Monserrat Bustelo: Inter-American Development Bank, 1300 New York Avenue, N.W., Washington, DC 20577. E-mail: MONSERRATB@iadb.org

Luca Flabbi, Department of Economics, University of North Carolina - Chapel Hill, 107 Gardner Hall, CB 3305 Chapel Hill, NC 27599-3305. E-mail: <u>luca.flabbi@unc.edu</u>

Mariana Viollaz, Department of Public Administration, CIDE, Carretera Mexico-Toluca 3655, Ciudad de Mexico and Centro de Estudios Distributivos, Laborales y Sociales (CEDLAS). E-mail: <u>mariana.viollaz@cide.edu</u> The paper also includes a Web Appendix available at

 $[\]label{eq:https://sites.google.com/site/lucaflabbi/home/research/BusteloFlabbiViollaz2019_OnlineAppendix .xlsx?attredirects=0\&d=1.$

1 Introduction

1.1 Motivation

Recent years have seen major changes in global labor markets. Technical innovation is leading to changes in industrial structure, workers' productivity, and the process of matching workers' skills with firms' technology. This is the result of the ever-greater expansion of a *digital economy*, in which old jobs are lost, new jobs are created, and even the same jobs are approached differently and require different competences. If old jobs are lost but new ones are created, then new opportunities may arise for workers who were previously at a disadvantage in the traditional economy.¹

In this paper, we focus on a specific set of workers who belong to such a group: women. Gender gaps in the labor market have been a significant and persistent empirical regularity in most countries around the world (Blau and Kahn, 2017). The digital economy affects this longstanding gap and may act to reduce or magnify it, depending on the skills that men and women can acquire and deliver in the labor market. Focusing on a set of high-income countries, OECD (2018a) finds that differences in skill endowments between male and female workers account for some but not all of the wage gap. When controlling for skills relevant to the digital economy, the gender wage gap is reduced by 1 to 12 percentage points.² OECD (2018b) also shows evidence of gender gaps in skills, tracing some back to human capital acquisition during the schooling years. However, among recent generations, the study finds that the gender gap in skills relevant to the digital economy is more ambiguous: In some areas, girls outperform boys (e.g., collaborative problem solving) but are outperformed by boys in others (e.g., digital-related skills).

Our report focuses on Latin American countries where no systematic analysis of the interaction between the digital economy and the gender gap has been undertaken.³ The primary reason for the

¹ A number of reports by multilateral organizations and international think-tanks are making a similar point: for the OECD, see OECD (2018a; 2018b); for the World Bank, see Box 2.10 in World Bank (2016); for the ILO, see Ryder (2018); for the WEF, see World Economic Forum (2018); for the IDB, see Busso et al. (2017).

² See Section 1.3 in OECD (2018a). The highest reduction is for Japan, which has the highest gender wage gap among OECD countries at about 40%.

³ Some country reports are starting to look at gender differences in skill acquisition, but without relating the results to gender differentials in the labor market. See, for example, OECD (2016) on Chile. The report – using PIAAC data –

absence of systematic contributions across Latin America relates to data limitations: Representative individual-level data, capable of capturing the skills required in the digital economy, are scarce. We are taking a first step in this analysis by using data sets comprising this relevant information for four Latin American countries: Chile, Bolivia, Colombia, and El Salvador. We have obtained the relevant information by combining two different surveys that collected data on skills: the *Programme for the International Assessment of Adult Competencies* (PIAAC) survey, available for Chile in 2014, and the *Skills Towards Employment and Productivity* (STEP) survey, available for Colombia and Bolivia in 2012 and for El Salvador in 2013.

After extracting the estimation sample, we document gender gaps in skills endowments and estimate returns to skills. We control for a variety of factors, including heterogeneous returns by workers' human capital and by sectors' digital intensity. We check the robustness of the results, with respect to the endogenous selection of women in the labor market. Finally, we decompose the gender gap in the portion resulting from differences in returns to skills and differences in skill endowments.

Two specific skills yield significant returns across the board: STEM (*science, technology, engineering, and math*) and ICT (*information and communication technologies*). ICT returns are similar across genders, with an increase of one standard deviation in the skill generating a return in hourly wages of about 17%. By contrast, STEM returns differ significantly between genders, with an increase of one standard deviation generating a 15% return for men and a 9% return for women. Results by country show some differences in magnitudes. Returns to STEM are higher than the aggregate in Chile and lower than the aggregate in Bolivia, Colombia, and El Salvador. We also estimate that higher levels of education imply significantly higher returns for men but not for women. The estimates indicate that STEM endowments and returns play an important role in explaining the aggregate gender gap in hourly wages; the lower endowments for women may be responsible for up to 30% of the gap, while the lower returns for women may account for close to 80%.

finds that "men outperform women in both literacy and numeracy. These gender gaps are among the widest observed across OECD countries" (page 6). Busso et al. (2017), on study skill formation in the Latin American and the Caribbean region, provided evidence on the availability and adoption of digital technologies but not on their relation to the gender gap.

1.2 Structure

The paper is organized as follows: Section 2 describes the data and the specific indicators we build to describe the digital economy. Section 3 provides standard descriptive statistics on the gender gaps in the labor market, integrating gaps in skills relevant to the digital economy. Section 4 provides estimates of the returns to these skills, according to gender. The estimates control for a rich set of individual characteristics and, in some specifications, for selection in the labor market. We also provide evidence regarding the interaction of the skills with the sectors most exposed to the digital economy, as well as with education levels. Section 5 focuses on the gender gap in hourly wages and unpacks the extent to which the gap can be accounted for by differences in returns, relative to differences in the endowment of digital-economy-relevant skills. Section 6 provides final comments and a summary of the results. Appendix A includes details about the construction of the variables of interest, Appendix B presents the complete regression results, and Appendix C provides results according to country. A Web Appendix includes additional specifications and robustness checks.

2 Data

We combine information from two surveys that collected data on skills at work. First, we use information accessible in the PIAAC survey, available for Chile in 2014. The PIAAC is a survey of adult skills, collected by the OECD, which measures adults' proficiency in key information-processing skills – literacy, numeracy, and problem solving – and gathers data on how adults use their skills at work. The survey is implemented by an international consortium, contracted by the OECD, and is designed to ensure international comparability of the data. Second, we use information from the STEP survey, available for Colombia and Bolivia in 2012 and El Salvador in 2013. STEP is a survey collected by the World Bank and gathers data on skills in urban areas of low- and middle-income countries. It also shares the main objectives and structure of the PIAAC questionnaire. It strives to record the data in a way that renders it comparable across different countries.

Still, two concerns remain regarding the comparability of STEP and PIAAC for our purposes. The first relates to differences in sampling: STEP is representative of people aged 15-64, living in urban

areas, while PIAAC is representative of people aged 16-65, living in both urban and rural areas. Due to STEP's urban focus, we exclude from the estimation sample workers employed in sectors represented more heavily in non-urban areas: agriculture, forestry and fishery, as well as mining and quarrying.⁴ Due to the slightly disparate age range between the two surveys, we include individuals aged 16-64 in the estimation sample. We also restrict the sample to wage employees of the private sector, except for El Salvador, where it is not possible to distinguish between the private and public sector. These restrictions generate a final estimation sample with 1,541 observation for Chile, 510 for Bolivia, 668 for Colombia, and 522 for El Salvador.

The second relates to differences in survey questions. PIAAC and STEP use different questionnaires to collect information. Still, the questions about labor market outcomes, demographic characteristics, socioeconomic variables, and ability measures generate variables that allow for a high degree of comparability between the surveys. The differences in survey questions become more relevant when crafting measures of skills, as eliciting skills is a more difficult, and arguably less agreed upon, task than eliciting demographic characteristics or standard labor market outcomes and, because skills are at the center of the paper's contribution. For these reasons, we devote the next section of the paper to describing how we build the relevant skills measures from the surveys' questions and implement a procedure to maximize comparability across countries and surveys.

Throughout the paper, the unit of observation of the data remains the individual worker. The level of analysis operates at both the country level and the aggregate level, pooling all countries into a single data set. Typically, we discuss the results from the pooled sample in the main text of the paper and report country-level results in the Web Appendix.

⁴ We could not limit our direct focus to the urban population in Chile because PIAAC does not report an indicator for urban and rural areas. The only geographic indicator reported by PIAAC is the OECD territorial level 2 region, which, for Chile, corresponds to 11 of the 15 regions of the country. We have combined this information with the rural/urban indicator contained in a nationally representative household survey for Chile – the 2013 CASEN survey. Based on CASEN, we can calculate the proportion of the population in each region living in urban areas. Results show that most of the population in each Chilean region is concentrated in urban centers. In most of the regions, more than 90% of the population lives in urban areas, with the most rural region (Maule) reporting 67% of urban population. Still, we have performed a robustness exercise, where we eliminated regions with less than 85% or 75% of the population living in urban areas. Our results were barely affected.

2.1 Skills measures

We build the skills measures to summarize the information collected in the numerous relevant survey questions and in order to maximize the comparability with previous literature and across countries. We focus on skills relevant in the digital economy. The previous literature has identified ICT and STEM quantitative skills as critical to thriving in the digital era (OECD, 2018a; OECD, 2018b). It has also argued that soft, social, and interpersonal skills, jointly with creativity and critical thinking, will become critical in future, innovative labor markets (Ryder, 2018; World Economic Forum, 2018; OECD, 2018b). To capture both sets of competencies, we follow the classification proposed by Grundke et al. (2017) in preparatory work for a series of OECD reports on skills and the digital economy.⁵ They define the following six sets of skills using data from the PIAAC survey on OECD countries:

1. *ICT skills* are those related to ICT use, from reading and writing emails to using Word and Excel or a programming language.

2. *Managerial and communication skills* capture strategic management abilities for the generation of sustainable competitive advantages, including human resource management practices, which encompass negotiating with people, instructing, advising, and persuading people.

3. *Readiness to learn and creative problem-solving skills* include the ability to solve problems in the process of developing organizational capabilities and improving firm performance.

4. *Self-organization skills* capture workers' flexibility to adapt to changes and absorb shocks.

5. *Marketing and accounting skills* include abilities associated with accounting tasks, such as reading financial invoices and bills, and with front-of-office marketing tasks, such as interacting with clients.

6. *STEM quantitative skills* include the ability to apply algebra, simple or complex, and prepare charts and tables.

In Table A1 in Appendix A, we provide details on the exact questions from PIAAC and STEP that we use to generate each skill measure and the metric associated with each variable. Variables from

⁵ See OECD (2018a, 2018b).

PIAAC derive exactly from the definition proposed by Grundke et al. (2017): We construct the same set of measures using the PIAAC for Chile as they did using the PIAAC on the 31 OECD countries in their analysis.⁶ In the STEP surveys for Bolivia, Colombia, and El Salvador, we identify the questions aimed at capturing the same skills and that are as close as possible to those available in PIAAC.

The first important difference between the two surveys is that most variables in PIAAC capture the frequency of doing certain task (from "never" to "every day") through categorical variables, while STEP involves a combination of binary questions (whether a person performs a certain task at work or not) with some categorical frequency questions as in PIAAC. In order to overcome these differences between surveys, we construct the skill measures applying a standardization methodology used by previous studies (Acemoglu and Autor, 2011; Almeida et al., 2018). First, for each variable associated to a skill category, we calculate the mean and standard deviation and standardize the variable by subtracting the mean and dividing by the standard deviation. Second, for each worker in the sample, we obtain a score for each skill category by adding all the standardized variables obtained in the first step. Third, for all skill measures to have a zero mean and standard deviation of one, we do one additional standardization of the six skill categories by subtracting their mean and dividing by their standard deviation. The standardization is performed for each country separately.

The second important difference concerns the difference in the actual questions. Many important questions are extremely similar, for example, those related to ICT skills or STEM skills, but there are others with more salient differences. For example, to build marketing and accounting skills, PIAAC includes a question on the frequency of selling products or services, but STEP does not. Our strategy here has been to base our choices on previous literature using the STEP survey to construct skills measures (Di Carlo et al., 2016; Roseth et al., 2016; Lo Bello et al., 2019) or combining both PIAAC and STEP to compare across countries (Lewandowski et al., 2019). The main difference between these previous studies and our paper is that they have focused on a different set of skills, mirroring the measures available in U.S. surveys instead of those available in the OECD PIAAC surveys.⁷

⁶ The only difference is the exclusion of a variable capturing physical skills that Grundke et al. (2017) included in the ICT measure with a negative weight.

⁷ They focus on the difference between cognitive and manual skills and between routine and non-routine tasks.

2.2 Digital intensity

In our econometric analysis, we also estimate differential returns, according to the digital intensity of each economic sector. Digital intensity refers to the degree in which the production and the performance of a given sector has been impacted by the digital transformation, i.e., the same set of technological innovations that have made relevant the skills that are the object of this study. The process of digital transformation is complex, involving more than a mere investment in computers, software, cold computing, or other digital tools, but also "embedding them in production with the appropriate human capital and using them when dealing with clients and suppliers" (Calvino et al., 2018). Looking at digital intensity is important in our context because the skills most relevant to the digital economy may be complementary to the technologies and business practices used by firms. While we do not have data granular enough to explore firms' characteristics, we can, at least, study the heterogeneity on returns by the degree of digital intensity at the sector level.

We proceed by defining an indicator variable of digital intensity for each of 36 economic sectors following the classification established by Calvino et al. (2018) for a subset of OECD countries. Using data from the Latin American region to define the digital intensive indicator variable would be better but lack of data prevents us to do that. Calvino et al. (2018)'s classification employs seven variables to describe a given sector as more or less digital-intensive: level of investment in ICT equipment, level of investment in software and databases, purchases of ICT services, purchases of ICT goods, robot use, revenues from online sales, and intensity of ICT specialists. None of these variables are reliably collected for Latin American and the Caribbean (LAC) countries, so we use the same indicator defined by Calvino et al. (2018) but apply it to our LAC countries.

To determine the extent to which transferring the OECD classifications to LAC countries represents a credible way to proceed, we use the only variable with some correlation to digital intensity available on the LAC region. The International Federation of Robotics (IFR) provides counts of the stock of robots by economic sector, country, and year worldwide, including the four countries under investigation here. We calculate the average number of robots in each economic sector for the entire region, taking the median value across sectors and classifying as "digital intensive" those sectors above the median. We compare this classification specification from the Latin American region to the one appearing in Calvino et al. (2018) to evaluate the viability of the

robot use component of their digital intensive indicator. The comparison indicates a correlation of 0.8 between the Latin American classification and the OECD classification, lending some weight to the use of the OECD classification in our estimates. Still, some important differences exist in the *intensity* of robot use between regions. Table A2 in Appendix A shows the average number of operational robots for seven aggregate economic sectors in the Latin American region, in the United States, and in Western Europe. The number of operational robots is between 4 and 100 times larger in the United States and Europe, relative to Latin America.

In conclusion, the comparison shows that LAC countries and high-income OECD countries are generally similar, in terms of the sectors of the economy with high digital intensity, but that the degree of intensity is likely to be much smaller in LAC countries than in OECD countries.

3 Descriptive Analysis: The Gender Gap

We first describe the gender gaps in the labor market and the gender differentials in the relevant skills variables. In terms of human capital measures, Table 1 documents the high rates of education of the women in our sample, relative to the men: 79% of women are high-educated (upper secondary education completed or more), while the percentage is 71% for men.

In terms of labor market performance, the usual gender gap in earnings emerges: The average wage of low-educated women is 83% of the average wage of low-educated men; for the high-educated group, the value is 87%. We also find the usual gender asymmetries in the distribution of employment, according to sectors and occupations. Services and sales is women's main occupation: 32% of women are employed in this occupational category, relative to 16% of men. Among men, crafts and related trades is the main occupation, employing about 22% of the workforce. In terms of sectors, both women and men are largely employed in the services sector, but the percentage for women is higher, reaching 93%, while, for men, it is 86%. A similar pattern is found when comparing the percentages of women and men employed in digital-intensive sectors. For women, we are looking at 53%, while, for men, it is 50%. The usual gender gap in labor market participation is also confirmed by our data: Figure 1 shows that, on average, 34% of women in the sample do not participate in the labor market, compared to 18% of men. There are

important differences by education levels: Low-educated women participate at a 54% rate, while high-educated women participate at a 73% rate.

Gender differences, with respect to skills relevant to the digital economy, are reported in Figure 2 and at the bottom of Table 1. The largest difference relates to STEM, where the average skill level for men is 0.12 of a standard deviation *above* the mean, while the average for women is 0.15 of a standard deviation *below* the mean.⁸ A smaller disadvantage for women is observed in the context of ICT skills, where they hold a value 0.03 of a standard deviation below the mean compared to a similar value ,which is, of course, above the mean for men. On all the other three skills, men and women do not show noticeable differences.

4 Econometric Analysis: Returns to Skills by Gender

This section presents estimates of the impact of skills on hourly wages, according to gender. We also analyze how these returns to skills interact with the digital intensity of the sector in which the worker is employed. The interaction is relevant to understand how skills play a role in the process of matching worker' skills with firms' technology. It also considers that while the digital economy is expanding, it may have not yet reached all the sectors of the labor market.

The econometric analysis focuses on estimating equations of the form:

$$lnw_{ij} = x_{ij}'\beta^g + \alpha^g s_{ij} + \gamma_j^g + \varepsilon_{ij}$$
(1)

where the natural log of the hourly wage of individual *i* of gender *g* in country *j* (w_{ij}) is regressed on a set of individual-level controls (the vector x_{ij}), a measure of the individual skill relevant in the digital economy (s_{ij}) and a country-fixed effect (γ_j). The coefficient of interest is α , representing the percentage change in earnings associated with a unit increase in the amount of skill *s*. Following the literature estimating earnings functions, we label this coefficient the *return to skills*.

⁸ All the skill indexes are normalized to be mean zero and a standard deviation of one in the overall population.

It is worth noting here that the parameters to be estimated are denoted by the superscript g, to indicate that we are estimating gender-specific coefficients. In other words, we are estimating equation (1) separately by gender.⁹

This specification is different, with respect to the one used by OECD (2018a), which performs a similar analysis on OECD countries. OECD's study estimates gender-specific returns to skills by interacting a gender dummy with the skill variable and then estimating on the pooled sample of men and women. In terms of equation (1), this is equivalent to having the superscript *g*, but only on the α coefficient. We prefer to let the estimator be flexible enough to capture gender-specific returns on all the variables. Without this flexibility, the returns to skills could capture some of the gender-specific impacts of the other covariates. We find that this specification is, indeed, relevant; numerous coefficients are significantly different between men and women. In Section 4.4, we show the extent to which using the OECD (2018a) specifications instead impacts our results.

The individual-level controls included in the vector x_{ij} are quite parsimonious, due partly to data availability and also to the returns we aim to identify. Specifically, the vector includes a constant, age and age squared, an indicator variable of completed upper secondary education or more, and literacy and numeracy test scores.¹⁰ In short, the baseline specification controls for broad human capital indicators (age and education), for a proxy of basic cognitive abilities (literacy and numeracy scores) and for aggregate differences between the economies (the country-fixed effects). The only additional controls available across all data sets, available for our inclusion in the specification, are sectors and occupations indicator variables. We decided not to include them in the baseline specification for the following reason: Choice of sector and occupation is endogenous and, as such, the overall returns to skills may legitimately include the return to the choice of a certain occupation/industry cell. Since we are interested in gender gaps in the overall return to skills, we have preferred not to decompose the returns in the component due to occupation and sector choices and in the component due to differences within a given sector and occupation cell.

 $^{^{9}}$ To simplify notation, we do not add the indicator g to the observed variables, but it is understood that we will include in each regression only individuals belonging of a specific gender.

¹⁰ Literacy and numeracy test scores are, unfortunately, not available for all countries. Literacy scores are available for Bolivia, Colombia, and Chile, while numeracy scores are only available for Chile. We have therefore decided to have them interact with the country's fixed effects in the section for specification, pooling all the countries together.

Still, estimating the coefficient conditional on industry and occupation is informative, and we present these estimates in Section 4.4.

4.1 Main Results

The main results on returns to skills are presented in Figure 3. The two main cognitive skills relevant in the digital economy – ICT and STEM – show positive and significant returns for both men and women, but, in STEM, the returns are significantly higher for men.¹¹ Specifically, a one standard deviation increase in the STEM skill measure implies a 15.1% increase in wages for men and an 8.7% increase in wages for women. ICT skill returns are more similar across genders, ranging from 16.9% for women to 17.6% for men. The returns to the other skills are lower but still positive and significant except for marketing and accounting for women.

Compared with the returns estimated in OECD countries, the returns to ICT and STEM are higher than the estimated average over the pooled set of 31 OECD countries. OECD (2018a) estimates returns to ICT skills to be about 12% for women and about 10.5% for men. The returns to STEM skills are estimated at 4% and 4.5%, for women and men, respectively. Contrary to the results for the four LAC countries, there is a gender gap in returns to ICT, favoring women, while there is no significant difference in returns to STEM by gender.¹²

We have tried to determine whether the difference in results, with respect to OECD (2018a), derives from differences in the specification. As mentioned above, even when the variables employed are comparable, the specifications are slightly different: OECD (2018a) estimates that by having a gender dummy interacted with skills, while we prefer a more flexible specification, where all the returns are allowed to be gender-specific. We have run the specification preferred by OECD (2018a) on our data and found that the difference in specification explain only a small portion of the difference. We discuss the results in Section 4.4.

The main results of the returns to skills interacted with the level of digital intensity in the sector are presented in Figure 4. The figure reports the additional impact of a given skill when the worker

¹¹ We report point estimates with the dot and the diamond, and the 90% confidence interval with vertical lines. As a result, any time that the vertical line does not cross zero, it means that the coefficient is significantly different from zero.

¹² See Figure 1.7, page 16, OECD (2018a).

is employed in a sector with high digital intensity. The exercise is meant to capture the extent to which a portion of the overall returns discussed above is due to composition effects across sectors, i.e., if the skills have significant and positive returns only in those sectors that are more impacted by the digital economy, or if they are equally valuable across all sectors. The figure shows that digital intensity does not generally have a significant impact: Most of the coefficients are not significantly different from zero, and the point estimates are very close to zero.¹³ Again, this result stands in stark contrast with the findings for OECD countries. OECD (2018a) estimates that working in digital-intensive industries increases returns to skill by about 2 to 3 percentage points.¹⁴ The results for our four Latin American countries indicate that either digital intensity is not very pronounced in Latin America or that the skills under consideration are valuable across all the sectors of Latin American economies. Even with the data at our disposal, we cannot disentangle the two explanations, but some of the evidence discussed in Section 2.2 indicate that digital intensity is, indeed, less pronounced in Latin America than in OECD countries. Specifically, Table A2 in Appendix A shows that the average number of operational robots for seven aggregate economic sectors (the only direct measure of digital intensity available for LAC countries) is between 4 and 100 times larger in the United States and Western Europe than in Latin America. This result is consistent with the analysis based on the broader Digital Adoption Index, reported by Busso et al. (2017). The index shows that Latin America lags behind high-income economies in terms of availability and adoption of digital technologies.¹⁵ In other words, if the sectors of the economy with high digital intensity are similar between LAC countries and high-income OECD countries, the extent of the intensity is much smaller in LAC.

To summarize the results obtained by aggregating the data gleaned from the four countries under study, (i) we estimated significant and large returns to ICT and STEM, with no significant gender differences on the first, but a significant gender gap in favor of men on the second. (ii) For the other skill measures, we found positive but lower returns with mainly not statistically significant differences between women and men. (iii) The digital intensity of the sectors does not seem a crucial factor in determining these returns.

¹⁴ See Figure 1.3, page 12, OECD (2018a).

¹³ The exception are self-organization and marketing & accounting, which have, respectively, slightly smaller and slightly higher returns when used in digital-intensive sectors.

¹⁵ See Figure 4.8, page 95, Busso et al. (2017).

In Appendix C, we look at each country in isolation from the others; we actually found results for each that are qualitatively similar to the aggregate for Chile, Bolivia, and Colombia but with some differences in magnitude. Returns to STEM are higher than the aggregate of Chile and lower than the aggregate of Bolivia and Colombia. El Salvador, on the other hand, does not report a gender gap favoring men in returns to STEM; both men and women have positive but similar returns. Returns to ICT in El Salvador are also a little different than the aggregate, indicating a higher (but not significantly so) return for women. Extra returns for working in digital-intensive sectors estimated by country confirm the aggregate results: Digital intensity does not significantly increase returns to skills.

4.2 Heterogeneous Effects

We focus on the heterogeneous effect on the individual characteristic most relevant with respect to skills formation and acquisition: schooling levels. We focus on upper secondary education, and we report the differential effect on the returns of individuals who completed this educational level with respect to individuals who completed less. The results are reported in Figure 5. They show that higher levels of education yield significantly higher returns for men but not for women. For women, the increase in returns never departs significantly from zero, and the point estimates are always lower than those for men. In the next section, we discuss whether these differences may be due to the different selection in the labor market by gender.

These results are more mixed when looking at regressions by country (Figure C4 in Appendix C). In Chile, we estimate significantly higher returns for high-educated men on all skills except readiness to learn. In Bolivia and El Salvador, only half of the skill types demonstrate significant increases; in Colombia, only one. What remains broadly similar to the aggregate results is that the increase in returns is higher for men than women. The only important exception is Colombia, where the point estimate of increase in ICT and STEM is higher for women. However, the estimated coefficient is not significantly different from zero in either case.

4.3 Selection in the Labor Market

Women participate in the labor market less than men. This is a common feature among most labor markets, and it holds in our samples. Figure 1 shows that women participate in the labor market less than men at all education levels. The Web Appendix shows the results of each country. Because participation is non-random – i.e., it is the result of a choice – the women we observe working may be systematically different from those not working. If the characteristics over which they are different are related to their performance in the labor market, then the return to skills we estimated are only representative of the population of women working and not of the entire population of women. This bias may affect some relevant policy implications. For example, if a policy is successful in increasing the skill levels of women, should we expect the women's returns that were affected by the policy to be more like those of the women currently participating or the ones who are not? Because some of the factors related to productivity remain unobserved, it is not easy to answer this question. It may therefore be useful to estimate a set of returns that can indicate if the estimated returns of women currently working are significantly different from those of the overall population of women.¹⁶

We adopt a standard Heckman selection model (Heckman, 1979) to obtain returns that consider selection in the labor market. To avoid relaying only on distributional assumptions for identification, we adopt an exclusion-restriction: We assume that each woman's number of children affects her participation decision but not the wage paid to her.¹⁷

Results are presented in Figure 6 and show relatively small changes with respect to the baseline specifications reported in Figures 3 and 5. Still, two small differences are worth mentioning. First, the point estimate of the return to ICT becomes higher for women than men. However, both in the baseline specification and in the selection-correction specification, the differences in returns between men and women are not significantly different from each other. Second, the gap in favor

¹⁶ Notice that in principle, the same selection problem also applies to men but because a high proportion of men participate in the labor market, we follow the common practice of ignoring selection in the labor market for them. In other words, even if not all men participate in the labor market, the small proportion of men who do not is unlikely to affect the overall estimates very much.

¹⁷ This assumption is common in the literature, but it is not without limitations. If it is true that firms cannot offer wage contracts contingent to the presence of children, some evidence shows that the unequal distribution of household work between men and women may generate a systematic (and potentially causal) association between wages and the presence of children.

of men in returns to STEM is reduced because the point estimate for women increases from 8.7% to 9.5%.

By looking at results by country (reported in Figure C5 of Appendix C), we can see that these minor changes are mainly driven by Chile. In Chile, controlling for selection increases women's returns to ICT from 18.9% to 24.0% and returns to STEM from 11.0% to 18.3%. Another relevant change takes place in Bolivia, where women's returns to STEM become significantly different from zero when controlling for selection.

With respect to heterogeneous returns, correcting for selection does not qualitatively change the results. The results are reported in Figure 7 and should be compared to Figures 4 and 5. Overall, they confirm the lack of significantly higher returns for agents working in the digital-intensive sector and reiterate that only men have significantly higher returns when they complete upper secondary education.

Finally, estimates obtained by correcting for selection show that this is not the source of differences with respect to returns estimated in OECD countries (OECD 2018a). Selection-correction slightly increases the differences because it leads to slightly higher returns to ICT and STEM for women.

4.4 Robustness Analysis

In this section, we provide a robustness analysis along three dimensions. First, we study how the results change when we replicate the specifications of the OECD (2018a); we constrain all the coefficients to be the same for men and women except for the returns to skills. Second, we study how the results change when we control for sectors and occupations. Finally, we check if the results controlling for selection are affected by using different exclusion-restrictions in the selection model.

Results constraining all the coefficients to be the same for men and women except for the returns to skills are reported in Table B7. This specification consists in running equation (1) on a sample pooling together men and women and then interacting the women indicator variable with the variable measuring the specific skill under consideration. Our favorite baseline specification allows all the coefficients to be gender-specific because it allows for more flexibility. This different constrained version lets only the return to skill be gender-specific; as a result, returns to skills may

capture some of the gender-specific impacts of the other covariates. By focusing on the two skills with the strongest impact in the baseline specification (ICT and STEM), the estimation results in the constrained specification are essentially identical to ICT and show a small difference to STEM. If, in the baseline specification, men's returns to STEM are estimated to be 15.2%, those for women are 6.7 percentage points lower. Men's returns on the constrained specification are estimated to be 14.9%, whereas those for women are 5.9 percentage points lower. This slightly lower gender gap in constrained specification is an indication that allowing for gender-specific returns on the other covariates and on country-fixed effects is useful for estimating returns closer to consistent values.

Results controlling for sectors and occupations are reported in Table B8. As mentioned, sectors and occupations are important controls given that each sector/occupation cell may require and use the skills in the digital economy differently. Conditioning on them may clarify if the differences in returns we observed are due to the sorting of men and women over these cells or to different returns within cells. Because occupational choices are endogenous and within-sector differences may be affected by unobservable characteristics, this analysis remains largely descriptive. In other words, the overall returns to skills may legitimately include the return to the choice of a certain occupation. The estimation results conditioning on sectors and occupation qualitatively confirm the main result in the baseline specification: We estimate significant returns to ICT and STEM, with no significant gender differences in the former but a significant gender gap in favor of men in the latter. Quantitatively, both ICT and STEM returns are estimated to be lower when sectors and occupation controls are introduced. For example, women's ICT returns decreased from 16.8% in the baseline specification to 7.0% in the specification controlling for sector and occupation; the corresponding values for men were 17.4% and 7.1%, respectively. These results indicate, probably not surprisingly, that a portion of the returns to ICT and STEM is because individuals who possess those skills work in a sector and occupation that pay relatively more in comparison to people with a lower level of these skills. More interestingly, the gender gap in ICT returns is unaffected when we change the specification, unlike the gender gap in STEM returns. In other words, conditioning on the sector and occupation cell, men's returns to STEM are estimated to be much greater than women's returns to STEM than in the economy as a whole. The results suggest that different choices of occupation and sector are not the main driver of the overall gender gap in STEM returns; women's occupational choices, vis-a-vis men's, decrease the gender gap in returns present within sector/occupation cells.

Results controlling for how different exclusion-restrictions affect our selection model are reported in Table B9. In our baseline selection model, we used the presence of children as the exclusionrestriction, a variable very likely to affect the participation decisions of women. However, not only fertility decisions are endogenous, but the presence of children may also directly impact wages and not only participation. Still, we are limited in terms of data availability on the exclusionrestrictions we can use in the selection model. Furthermore, the number of children is the only variable available on all data sets. However, if we limit ourselves to the STEP countries, we can use additional information. Specifically, we use information about whether the mother or father of the person answering the survey lives in the household. This presence may be very influential on the labor market participation decision of women, either because they can help in household production or because they need resources for their care. We therefore built an indicator variable for whether the mother or father of the person answering the survey is living in the household. We use it as an exclusion-restriction alone and jointly with the presence-of-children variable. Estimation results show very little sensitivity to the various exclusion-restrictions. For example, the overall STEM return in STEP countries is estimated to be about 5.7% without controlling for selection. When controlling for selection, it is estimated to be just a little lower, ranging from 5.13% when using only the presence of children as an exclusion-restriction, to 5.17% when using only the presence of grandparents, to 5.16% when using both.¹⁸

5 Post-Estimation Analysis: Decomposing the Gender Gap

The previous section presented estimates of the returns to skills and to other individual characteristics by gender. Using these estimates, it is possible to decompose the overall gender gap

¹⁸ We have also run OLS regressions using the exclusion-restriction variables directly in the wage regressions. We found that they are typically not significant, while they always are in the participation-decision equation. Although the exclusion-restrictions are not formally testable since we use them for identification, this reduced form results lends some credibility to them.

in the portion due to differences in returns and to differences in endowments. The most popular procedure to accomplish this is labeled in the literature as Oaxaca-Blinder decomposition.¹⁹ It is simply an accounting exercise based on mean differences. The regressions we have presented in the previous sections and defined in equation (1) are ordinary least squares (OLS) regressions guaranteeing that:

$$\overline{lnw_g} = \overline{x_g}'\hat{\beta}^g + \hat{\alpha}^g \overline{s_g} + \overline{\hat{\gamma}}^g \tag{2}$$

where the overbar denotes sample mean, the hat denotes estimated value, and the indicator g=m,w denotes gender-specific values. The country-fixed effect component is averaged out over all the individuals living in the country.²⁰ In other words, equation (2) states that OLS estimation guarantees that the mean of the dependent variable is exactly predicted by the estimated coefficients applied to the sample mean of the regressors. Based on this result and one step of algebra, the gender earnings gap – the difference in average earnings between men and women – can be decomposed in the portion due to differences in the sample mean of the regressors (the *endowments*) and in the portion due to differences in the estimated coefficients (the *returns*). Focusing on the returns and endowments of relevant skills related to the digital economy – denoted with s – we can write:

$$\overline{\ln w_m} - \overline{\ln w_w} = \hat{\alpha}^m (\bar{s}_m - \bar{s}_w) + (\hat{\alpha}^m - \hat{\alpha}^w) \bar{s}_w + (\bar{x}_m' \hat{\beta}^m - \bar{x}_w' \hat{\beta}^w) + (\bar{\gamma}^m - \bar{\gamma}^w)$$
(3)

To formally define it, we need to add an indicator function
$$d_i(j) = l$$
 if individual i lives in country j, leading to:

$$\bar{\hat{\gamma}}^g = \sum_{j=1}^J \sum_{i=1}^N d_i(j) \hat{\gamma}_j^g$$

¹⁹ See Blinder (1973) and Oaxaca (1973). For a more recent assessment of the approach, see Oaxaca and Ransom (1994). For a more general decomposition involving the entire distribution, see Machado and Mata (2005).

where the first term on the right-hand side denotes gender differences in endowments in digitaleconomy-relevant skills (weighted by the male return $\hat{\alpha}^m$) and the second term denotes gender differences in returns to those skills (weighted by the female endowment \bar{s}_w).

Figure 8 reports the ratio between each of these two terms and the overall gender gap. For example, looking at STEM skills, we estimate that both gender differences in endowments and in returns have a significant impact in explaining the gender gap. The point estimates indicate that endowments could be responsible for up to 30% of the difference and returns for up to 80%.²¹ We see this as the main result to be drawn from the decomposition. Two other significant effects are present: Differences in ICT endowments have a marginally significant impact, whereas differences in returns in marketing and accounting have a more substantial impact, with the potential to explain up to 65% of the gap. We perform an additional exercise by reporting the endowment and returns components when including all skills measures at the same time. Results appear in the last panel of Figure 8 and show that endowments explain approximately 40% of the gender wage gap, while the returns to skills do not have any explanatory power.

Results by country are reported in Appendix C, Figure C7. They confirm the overall main results for Bolivia, Chile, and Colombia: STEM has a significant impact, with differences in returns contributing up to 64% of the gap in Bolivia, up to 28% in Chile, and up to 90% in Colombia. El Salvador is the only country where STEM does not have either an economically significant or a statistically significant impact on the earnings gender gap. In terms of other skills, the country-by-country results show that the significant impact of marketing and accounting in the aggregate sample was mainly driven by Colombia.

6 Conclusion

We studied gender differentials in the labor market with a focus on skills relevant to the digital economy. Specifically, we estimated the returns to skills using standard wage regressions, but we allowed for heterogeneous effects and considered selection in the labor market. We focused on

²¹ Recall that the estimated returns are based on running separate regressions, each with only one of the digitaleconomy-relevant skills at the time. Since there are likely to be correlations between skills, the estimates report the maximum impact that the skill may have in explaining the gap.

four Latin American countries for which all the relevant data are available: Bolivia, Chile, Colombia, and El Salvador.

We found that the group of skills described as ICT and as STEM have significant positive returns for both men and women. However, there is a significant gender gap in favor of men on the returns to STEM, while the returns to ICT are very similar between the two genders. We also estimated the returns in specifications allowing for heterogeneity by education and sector of occupation. We found that having upper secondary education or higher implies significantly higher returns for men but not for women. Unlike the results obtained on OECD countries (OECD, 2018a; OECD, 2018b), we did not find any significant impact related to working in a digital-intensive sector. This means that either digital intensity is less pronounced in Latin American countries than OECD countries or that the skills under consideration are valuable across all sectors of Latin American economies. All these results are confirmed when controlling for selection in the labor market for women. We also performed a robustness analysis where we studied how results change by varying the specification and by employing various exclusion-restrictions in the selection model.

Finally, we provided a decomposition of how skills relevant to the digital economy may contribute to the gender gap in wages. By computing a standard Oaxaca-Blinder decomposition based on our estimates, we concluded that both gender differences in endowments and in returns to STEM skills have a significant impact in explaining the gender gap in earnings. The point estimates indicate that endowments could be responsible for up to 30% of the difference and returns for up to 80%.

The summary of results provided so far is aggregated over the four countries under consideration. Many of our results were confirmed when performing the same analysis for each country separately. The most relevant differences between countries concerned returns to STEM. They were estimated to be higher than the aggregate in Chile and lower than the aggregate in Bolivia and Colombia. El Salvador is the only country not reporting a gender gap in favor of men in returns to STEM. As a result, El Salvador is the only country for which gender differences related to STEM skills do not explain the overall gender gap in earnings.

Why returns to STEM exhibit such strong gender asymmetries can only be partially answered by our analysis. The list of possible explanations is broad. The first possibility is selection into different sectors and occupations; men and women may systematically choose to work in different jobs, even when possessing similar skills. Our robustness analysis controlling for sectors and

occupations indicated that this is not the case: The within-sector and occupation gaps are higher than the overall gap. The second possibility is selection in the labor market, i.e., women participate in the labor market at a lower proportion than men. If only the relatively less productive women participate, then their lower returns may reflect a genuine productivity differential. Our analysis estimating a series of selection models indicated that this mechanism may at most explain a very small portion of the differential. The third possibility is different labor market experiences and labor market attachments. Some evidence suggests that time investments in household production are very different between men and women.²² This is reflected both in the extensive and intensive margins of labor supply. The extensive margin differential is reflected in the lower participation rate for women, and the selection models we estimated take this into account. The intensive margin is reflected in hours worked, which is accounted for by using hourly wages as dependent variables in our estimations. However, there is an important dynamic element that our static estimates could not properly consider; over time, a lower labor supply implies lower accumulation of human capital on the job. Our controls for labor market experience is very crude, and therefore, some of the impact of human capital accumulation on the job may be picked up by other returns, such as returns to education and returns to skills. The only objection to this explanation is that human capital accumulation on the job should involve all skills, not only STEM skills, while we estimate the gender gap only on STEM returns. The fourth possibility is discrimination against women in STEM professions. There is a growing, even if by no means conclusive, literature suggesting that STEM and related professions may be more prone to gender discrimination than other professions.²³ With the data at our disposal, we neither can exclude nor confirm such an explanation. Finally, the fifth possibility is that the quality of the STEM skills acquired by men and women may be different, explaining the gap in terms of underlining productivity differentials. This may result from human capital acquired before entering the labor market. By controlling for numeracy score, our regressions should condition on some of those differences, but the difference may result also from human capital acquired on the job after entering the labor market. This is the same mechanism we described when discussing the dynamic implication of gender differences in

²² Evidence collected in time-use survey provides convincing evidence in this respect. For cross-country evidence, see van der Lippe et al. (2011) and Apps and Rees (2005). For evidence on a Latin American country, see Salazar-Saenz (2019) on Colombia.

²³ See, for example, Buffington et al. (2016) and Carrell et al. (2010).

labor supply. Given the limited controls for dynamic behavior in our empirical models, this is another mechanism that cannot be ruled out.

In conclusion, we found that if investing in skills relevant to the digital economy, specifically in ICT and STEM, may be beneficial for the labor market performance of both men and women, it remains a relevant open question in understanding why returns to STEM exhibit such a strong gender gap. Our results can rule out some explanations but cannot precisely point to the actual mechanism. At least two explanations remain good candidates to explain the gender gap in STEM returns: the presence of stronger discrimination against women in STEM-related professions than in other professions and the life-cycle impact of a different rate of human capital accumulation on the job between men and women. The first suggests policies directly targeting the labor market, and the second suggests policies improving work-life balance for both men and women.²⁴

²⁴ For a recent assessment of these policies in the U.S., see Doran et al. (2018).

7. References

- Acemoglu, D., & Autor, D. (2011). Skills, tasks and technologies: Implications for employment and earnings. In *Handbook of labor economics* (Vol. 4, pp. 1043–1171), Elsevier.
- Almeida, R. K., Fernandes, A. M., & Viollaz, M. (in press). Software adoption, employment composition, and the skill content of occupations in Chilean firms. *The Journal of Development Studies*.
- Apps, P., & Rees, R. (2005). Gender, time use, and public policy over the life cycle. *Oxford Review* of *Economic Policy*, *21*(3), 439–461.
- Blau, F. D., & Kahn, L. M. (2017). The gender wage gap: Extent, trends, and explanations. *Journal* of *Economic Literature*, 55(3), 789–865.
- Blinder, A. S. (1973). Wage discrimination: Reduced form and structural estimates. *Journal of Human Resources*, 8(4), 436–455.
- Buffington, C., Cerf, B., Jones, C., & Weinberg, B. (2016). STEM training and early career outcomes of female and male graduate students: Evidence from UMETRICS data linked to the 2010 census. *American Economic Review*, *106*(5), 333–338.
- Busso, M., Cristia, J., Hincapié, D., Messina, J., & Ripani, L. (2017). *Learning better: Public policy for skills development*. Washington, DC: Inter-American Development Bank.
- Calvino, F., Criscuolo, C., Marcolin, L., & Squicciarini, M. (2018). A taxonomy of digital intensive sectors (OECD Science, Technology and Industry Working Papers 2018/14). Paris, France: OECD Publishing.
- Carrell, S., Page, M., & West, J. (2010). Sex and science: How professor gender perpetuates the gender gap. *The Quarterly Journal of Economics*, *125*(3), 1101–1144.
- Di Carlo, E., Lo Bello, S., Monroy-Taborda, S., Oviedo, A. M., Sanchez-Puerta, M. L., & Santos,
 I. (2016). *The skill content of occupations across low and middle income countries: Evidence from harmonized data* (IZA Discussion Paper 10224). Bonn, Germany: IZA Institute of Labor Economics.
- Doran, E. L., Bartel, A. P., & Waldfogel, J. (2018, Month). Gender in the labor market: The role of equal opportunity and family-friendly policies (NBER Working Paper No. 25378). Cambridge. USA: National Bureau of Economic Research.
- Grundke, R., Jamet, S., Kalamova, M., Keslair, F., & Squicciarini, M. (2017). Skills and global value chains: A characterization (OECD Science, Technology and Industry Working

Papers 2017/05). Paris, France: OECD Publishing.

- Heckman, J. J. (1979). Sample selection bias as a specification error. *Econometrica*, 47(1), 153–161.
- Lewandowski, P., Park, A., Hardy, W., & Du, Y. (2019). Technology, skills, and globalization: Explaining international differences in routine and nonroutine work using survey data (IZA Discussion Paper 12339). Bonn, Germany: IZA Institute of Labor Economics.
- Lo Bello, S., Sanchez-Puerta, M. L., & Winkler, H. (2019). From Ghana to America: The skill content of jobs and economic development (Policy Research Working Paper 8758). Washington, DC: The World Bank.
- Oaxaca, R. (1973). Male-female wage differentials in urban labor markets. *International Economic Review*, 14(3), 693–709.
- Oaxaca, R. L., & Ransom, M. R. (1994). On discrimination and the decomposition of wage differentials. *Journal of Econometrics*, *61*, 5–21.
- OECD. (2016). Country note Chile: Skills matter: Further results from the survey of adult skills. Paris, France: OECD Publishing.
- OECD. (2018a) Towards the implementation of the G20 roadmap for digitalisation: Skills, business dynamics and competition. Paris, France: OECD Publishing.
- OECD. (2018b) Bridging the digital gender divide: Include, upskill, innovate. Paris, France: OECD Publishing.
- Roseth, V. V., Valerio, A., & Gutierrez, M. (2016). Skills and labor market outcomes: Results from large-scale adult skills surveys in urban areas in 12 countries. Washington, DC: The World Bank Publications.
- Ryder, G. (2018). As tech disrupts our jobs, it's not too late to turn pain into gain. Retrieved from https://www.weforum.org/agenda/2018/10/tech-disrupt-jobs-fourth-industrial-revolution-ilo/
- Salazar-Saenz, M. (2019). Household search model of the labor market with home production. mimeo, University of North Carolina at Chapel Hill.
- van der Lippe, T., de Ruijter, J., de Ruijter, E., & Raub, W. (2011). Persistent inequalities in time use between men and women: A detailed look at the influence of economic circumstances, policies, and culture. *European Sociological Review*, *27*(2), 164–179.

World Bank Group. (2016). World development report 2016: Digital dividends. Washington, DC:

The World Bank Publications.

World Economic Forum. (2018). *The future of jobs report*. Cologny/Geneva, Switzerland: Centre for the New Economy and Society.

Table 1: Descriptive Statistics

Low High- educated Low High- educated High- educated Educated educated educated educated educated Share of women 0.44 0.36 0.47 .
number odd Share of women (0.5) (0.48) (0.5) (0.48) (0.5) 34.67 37.05 34.05 35.18 37.39 34.28 Age (11.79) (13.43) (11.45) (11.45) (13.04) (10.25) (13.65) (11.25) (13.65) (11.25) (13.65) (11.25) (13.65) (11.25) (13.65) (11.72) (0.43) (0.41) (0.45) (0.27) (0.26) (0.27) (0.26) (0.27) (0.26) (0.27) (0.26) (0.27) (0.26) (0.27) (0.26) (0.27) (0.26) (0.5)
Share of women 0.44 0.36 0.47 V V Age 34.96 37.26 34.17 34.67 37.05 34.05 35.18 37.39 34.28 (11,79) (13.43) (11.08) (11.45) (13.04) (10.20) (12.05) (13.65) (11.22) Share of low-educated 0.21 0.21 (0.43) (0.41) (0.41) (0.45) (0.41) (0.45) (0.41) (0.45) (0.41) (0.45) (0.41) (0.45) (0.41) (0.45) (0.41) (0.45) (0.46) (0.45) (0.46) (0.45) (0.46) (0.45) (0.46) (0.45) (0.46) (0.45) (0.46) (0.46) (0.46) (0.46) (0.46) (0.46) (0.46) (0.46) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) (0.46) (0.47) <t< th=""></t<>
Age (0.5) (0.49) (0.5) 44.67 37.05 34.26 Age 34.69 37.26 34.17 34.67 37.05 34.05 37.39 34.28 Share of low-educated (0.25) (0.14) (11.45)<
Age 34,96 37,26 34,17 34,67 37,05 34,05 35,18 37,39 34,28 (11,79) (13,43) (11,08) (11,45) (13,04) (10,25) (12,05) (13,65) (11,22) Share of high-educated 0,25 0,21 0,24 0,24 0,24 0,24 0,25 Share of high-educated 0,75 0,76 0,76 0,77 0,79 0,77 0,79 Literacy score 0,79 0,76 0,80 0,79 0,74 0,80 0,79 0,77 0,79 Houty wage in dollars at PPP of 2005 4,55 3,02 5,07 4,28 2,68 4,70 3,21 5,39 Distribution of employment by occupation (1,61) (0,05) (0,18) (0,14) 0,08 (0,16) (0,17) (0) (0,2) Professionals 0,12 0,03 0,03 (0,33) (0,33) (0,31) 0,36 (0,27) (0,06) (0,31) Technicians and associate professionals
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $
Share of low-educated 0.25 0.21 0.29 (0.43) (0.41) (0.45) (0.41) (0.45) Share of high-educated 0.75 0.79 0.74 0.80 0.79 0.71 Literacy score (0.43) (0.41) (0.45) (0.26) (0.27) (5.6) Hourly wage in dollars at PPP of 2005 4.55 3.02 5.07 4.28 2.68 4.70 4.76 3.21 5.39 Managers 0.02 0.00 0.03 0.02 0.01 0.02 0.01 0.02 0.01 0.02 0.01 0.02 0.03 0.03 0.02 0.03 0.33 0.016 0.03 0.02 0.03 0.017
(0.43) (0.43) (0.41) (0.45) Share of high-educated 0.75 0.79 0.74 (0.45) Literacy score 0.79 0.76 0.80 0.79 0.74 0.80 0.79 0.77 0.79 Houty wage in dollars at PPP of 2005 4.55 3.02 5.07 4.28 2.68 4.70 4.76 3.21 5.39 Distribution of employment by occupation (4.72) 0.02 0.00 0.03 0.02 0.01 0.02 0.03 0.02 0.01 0.02 0.03 0.02 0.16 (1.64) (0.65) Professionals 0.10 0.01 0.13 0.12 0.02 0.16 0.17 (0.16) (0.17) (0.06) (0.18) Clerical support workers 0.12 0.03 0.16 0.13 0.31 0.31 0.33 (0.31) (0.33) (0.33) (0.31) 0.32 0.39 (0.33) (0.31) (0.32) (0.31) (0.35) (0.22) (0.33)
Share of high-educated 0.75 0.74 0.71 Literacy score 0.79 0.76 0.80 0.79 0.74 0.80 0.79 0.77 0.79 Luteracy score 0.79 0.76 0.80 0.79 0.74 0.80 0.79 0.77 0.79 Hourly wage in dollars at PPP of 2005 4.55 3.02 5.07 4.28 2.68 4.70 4.76 3.21 5.39 Distribution of employment by occupation (4.72) (3.34) (5) (4.59) (3.27) (4.79) (4.81) (3.37) (5.16) Managers 0.02 0.00 0.03 0.02 0.01 0.02 0.03 0.00 0.11 Professionals 0.10 0.13 0.12 0.02 0.16 0.13 0.11 0.17 (0.06) (0.31) Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.05 0.15 Clerical support workers 0.23 <t< td=""></t<>
(0.43) (0.41) (0.41) (0.45) Literacy score (0.79 0.76 0.80 0.79 0.77 0.77 0.79 Houly wage in dollars at PPP of 2005 4.55 3.02 5.07 4.28 2.68 4.70 4.76 3.21 5.39 Distribution of employment by occupation (4.72) (3.34) (5) (4.59) (3.27) (4.79) (4.81) (3.37) (5.16) Distribution of employment by occupation (0.66) (0.18) (0.14) (0.08) (0.16) (0.17) (0) (0.2) Professionals 0.10 0.01 0.13 0.12 0.02 0.15 0.08 0.00 0.11 Technicians and associate professionals 0.12 0.03 (0.33) (0.31) (0.31) (0.31) (0.32) (0.19) (0.32) Clerical support workers 0.14 0.55 0.17 0.16 0.66 0.19 0.12 0.03 0.17 (0.32) (0.31) (0.31) (0.31) (0.3
Llteracy score 0.79 0.76 0.80 0.79 0.74 0.80 0.79 0.77 0.79 (0.26) (0.28) (0.25) (0.25) (0.26) (0.27) (0.26) (0.27) (0.26) (0.27) (0.26) (0.27) (0.26) (0.21) (0.26) (0.21) (0.26) (0.21) (0.26) (0.21) (0.26) (0.21) (0.26) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) (0.21) <t< td=""></t<>
(0.26) (0.26) (0.26) (0.26) (0.26) (0.26) (0.26) (0.27) (0.26) Hourly wage in dollars at PPP of 2005 4.55 3.02 5.77 4.28 2.68 4.70 4.76 3.21 5.39 Distribution of employment by occupation
Houry wage in dollars at PPP of 2005 4.55 3.02 5.07 4.28 2.68 4.70 4.76 3.21 5.39 Distribution of employment by occupation (4.72) (3.34) (5) (4.59) (3.27) (4.79) (4.81) (3.37) (5.16) Distribution of employment by occupation 0.02 0.00 0.03 0.02 0.01 0.02 0.03 0.00 0.04 Professionals 0.10 0.01 0.13 0.12 0.02 0.15 0.08 0.00 0.11 Technicians and associate professionals 0.12 0.03 0.16 0.13 0.01 0.17 0.11 0.04 0.15 Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.03 0.16 0.13 0.12 0.03 0.16 0.6 0.19 0.12 0.05 0.15 Clerical support workers 0.23 0.22 0.33 0.14 0.42 0.14 0.42 0.14 0.23 0.33 0.14 0.23 0.33 0.16 0.16 <th< td=""></th<>
(4.72) (3.34) (5) (4.59) (4.79) (4.81) (3.37) (5.16) Distribution of employment by occupation 0.02 0.00 0.03 0.02 0.01 0.02 0.03 0.00 0.04 Managers (0.16) (0.05) (0.18) (0.14) (0.08) (0.16) (0.17) (0) (0.2) Professionals (0.3) (0.09) (0.33) (0.33) (0.3) (0.36) (0.27) (0.06) (0.31) Technicians and associate professionals 0.12 0.03 0.16 0.13 0.01 0.17 0.11 0.04 0.15 Clerical support workers 0.14 0.05 0.17 0.16 0.66 0.19 0.12 0.05 0.15 Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.616 0.16
Distribution of employment by occupation Nanagers 0.02 0.01 0.02 0.01 0.02 0.03 0.00 0.04 Professionals 0.10 0.01 0.13 0.12 0.02 0.15 0.08 0.00 0.11 Technicians and associate professionals 0.12 0.03 0.01 0.13 0.12 0.02 0.15 0.08 0.00 0.11 Technicians and associate professionals 0.12 0.03 0.16 0.13 0.01 0.17 0.14 0.05 0.17 0.16 0.057 (0.32) (0.19) (0.35) Clerical support workers 0.14 0.05 0.17 0.16 0.66 0.19 0.12 0.05 0.15 Services and sales workers 0.23 0.24 0.23 0.22 0.39 0.30 0.16 0.16 0.16 (0.42) (0.43) (0.42) (0.47) (0.49) (0.46) (0.37) (0.36) (0.27) (0.30 0.22 0.33 0.14 </td
Maragers 0.02 0.00 0.02 0.01 0.02 0.01 0.02 0.03 0.00 0.04 Professionals 0.10 0.01 0.11 0.12 0.02 0.16 (0.17) (0) (0.2) Professionals 0.12 0.03 (0.09) (0.33) (0.33) (0.13) (0.36) (0.27) (0.06) (0.31) Technicians and associate professionals 0.12 0.03 0.16 0.13 0.01 0.17 0.11 0.04 0.15 Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.05 0.15 Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 Craft and related trades workers 0.14 0.24 0.10 0.44 0.29 0.33 0.22 0.33 0.17 Craft and related trades workers 0.14 0.27 0.10 0.44 0.29 0
(0.16) (0.05) (0.18) (0.14) (0.08) (0.16) (0.17) (0) (0.2) Professionals 0.10 0.01 0.13 0.12 0.02 0.15 0.08 0.00 0.11 (0.3) (0.09) (0.33) (0.13) 0.13 0.11 0.13 0.01 0.17 0.11 0.04 0.15 (0.33) (0.17) (0.36) (0.34) (0.1) (0.37) (0.32) (0.19) (0.35) Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.05 0.15 Clerical support workers 0.13 0.22 0.38 (0.37) (0.23) (0.39) (0.33) (0.21) (0.36) Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 Craft and related trades workers 0.14 0.24 0.10 0.04 0.99 0.33 0.22 0.33 0.1
Protessionals 0.10 0.01 0.03 0.12 0.02 0.15 0.08 0.00 0.11 (0.3) (0.09) (0.33) (0.33) (0.13) (0.36) (0.27) (0.06) (0.31) Technicians and associate professionals 0.12 0.03 0.16 0.13 0.01 0.17 0.11 0.04 0.15 Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.05 0.15 Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 Craft and related trades workers 0.14 0.24 0.03 (0.28) (0.37) (0.38) (0.37) (0.36) (0.37) Craft and related trades workers 0.14 0.24 0.10 0.04 0.99 0.03 0.22 0.33 0.17 Clark and related trades workers 0.14 0.24 0.10 0.04 0.99 0.03 0.22 0.33 0.17 Plant and machine operators and assemblers 0.16 0.08 </td
(0.3) (0.09) (0.33) (0.13) (0.36) (0.27) (0.06) (0.31) Technicians and associate professionals 0.12 0.03 0.16 0.13 0.01 0.17 0.11 0.04 0.15 (0.33) (0.17) (0.36) (0.34) (0.1) (0.37) (0.32) (0.19) (0.35) Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.05 0.15 (0.35) (0.22) (0.38) (0.37) (0.23) (0.39) (0.33) (0.21) (0.36) Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.17 (0.34) (0.42) (0.47) (0.49) (0.46) (0.37) (0.36) (0.37) Craft and related trades workers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.17 (0.38) Plant and machine operators and assemblers 0.10 0.16
Technicians and associate professionals 0.12 0.03 0.16 0.13 0.01 0.17 0.11 0.04 0.15 (0.33) (0.17) (0.36) (0.34) (0.1) (0.37) (0.32) (0.19) (0.35) Clerical support workers (0.35) (0.22) (0.38) (0.37) (0.23) (0.39) (0.33) (0.21) (0.36) Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 (0.42) (0.43) (0.42) (0.47) (0.49) (0.46) (0.37) (0.38) (0.37) Craft and related trades workers 0.14 0.24 0.10 0.04 0.09 0.03 0.22 0.33 0.17 Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.19 0.14 0.19 0.12 0.33 0.11 0.12 0.11 0.15 0.33 0.10 0.14 0.13 <
(0.33) (0.17) (0.36) (0.34) (0.1) (0.37) (0.32) (0.19) (0.35) Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.05 0.15 Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 0.16 Craft and related trades workers 0.14 0.24 0.10 0.04 0.99 0.03 0.22 0.33 0.16 0.16 0.16 Craft and related trades workers 0.14 0.24 0.10 0.04 0.99 0.03 0.22 0.33 0.17 (0.34) (0.43) (0.3) (0.2) (0.28) (0.16) (0.41) (0.47) (0.38) Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.12 0.03 0.31 0.14 0.22 0.31 0.14 0.23 0.10 0.14 0.21 0.21 0.33 0.10 0.14 0.23 0.10 0.14
Clerical support workers 0.14 0.05 0.17 0.16 0.06 0.19 0.12 0.05 0.15 Services and sales workers 0.23 0.24 0.23 0.37 (0.23) (0.39) (0.33) (0.21) (0.36) Craft and related trades workers 0.14 0.24 0.10 0.04 0.09 0.03 0.22 0.33 0.17 Craft and related trades workers 0.14 0.24 0.10 0.04 0.09 0.03 0.22 0.33 0.17 (0.34) (0.43) (0.3) (0.2) (0.28) (0.16) (0.41) (0.47) (0.38) Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.19 (0.35) (0.4) (0.33) Elementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 Distribution of employment by economic sector (0.32) (0.31) (0.26) (0.28) (0.26)
(0.35) (0.22) (0.38) (0.37) (0.23) (0.39) (0.33) (0.21) (0.36) Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.07 (0.37) (0.34) (0.42) (0.41) (0.42) (0.41) 0.040 0.09 0.03 0.22 0.33 0.17 (0.34) (0.43) (0.3) (0.2) (0.28) (0.16) (0.41) (0.47) (0.38) Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.19 0.12 (0.3) (0.37) (0.28) (0.22) (0.31) (0.19) (0.35) (0.4) (0.33) Elementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 Distribution of employment by economic sector <t< td=""></t<>
Services and sales workers 0.23 0.24 0.23 0.32 0.39 0.30 0.16 0.16 0.16 0.16 (0.42) (0.43) (0.42) (0.43) (0.42) (0.47) (0.49) (0.46) (0.37) (0.36) (0.37) Craft and related trades workers 0.14 0.24 0.10 0.04 0.09 0.03 0.22 0.33 0.17 (0.34) (0.43) (0.3) (0.2) (0.28) (0.16) (0.41) (0.47) (0.38) Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.19 0.12 (0.3) (0.37) (0.28) (0.22) (0.31) (0.19) (0.35) (0.4) (0.33) Elementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 Distribution of employment by economic sector (0.32) (0.31) (0.26) (0.28) (0.26) (0.35)<
(0.42) (0.43) (0.42) (0.47) (0.49) (0.46) (0.37) (0.36) (0.37) Craft and related trades workers 0.14 0.24 0.10 0.04 0.09 0.03 0.22 0.33 0.17 (0.34) (0.43) (0.3) (0.2) (0.28) (0.16) (0.41) (0.47) (0.38) Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.19 0.12 (0.3) (0.37) (0.28) (0.22) (0.31) (0.19) (0.35) (0.4) (0.33) Elementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 (0.35) (0.44) (0.3) (0.35) (0.47) (0.3) (0.34) (0.42) (0.29) Distribution of employment by economic sector (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86
Crait and related trades workers 0.14 0.24 0.10 0.04 0.09 0.03 0.22 0.33 0.17 (0.34) (0.43) (0.3) (0.2) (0.28) (0.16) (0.41) (0.47) (0.38) Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.19 0.12 (0.3) (0.37) (0.28) (0.22) (0.31) (0.19) (0.35) (0.41) (0.41) (0.41) 0.12 (0.3) (0.37) (0.28) (0.22) (0.31) (0.19) (0.35) (0.4) (0.33) Elementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 Distribution of employment by economic sector (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87
Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.19 0.12 Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.19 0.12 (0.33) (0.37) (0.28) (0.22) (0.31) (0.19) (0.35) (0.41) 0.14 0.19 0.12 (0.35) (0.44) (0.3) (0.35) (0.47) (0.33) 0.10 0.14 0.23 0.10 (0.35) (0.44) (0.3) (0.35) (0.47) (0.3) (0.34) (0.42) (0.29) Distribution of employment by economic sector
Plant and machine operators and assemblers 0.10 0.16 0.08 0.05 0.10 0.04 0.14 0.19 0.12 (0.3) (0.37) (0.28) (0.22) (0.31) (0.19) (0.35) (0.4) (0.33) Elementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 Distribution of employment by economic sector Image: Construction of the sector Image: Construction
(0.3) (0.37) (0.28) (0.31) (0.31) (0.19) (0.35) (0.4) (0.33) Elementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 (0.35) (0.4) (0.3) (0.4) (0.3) (0.47) (0.3) (0.34) (0.29) Distribution of employment by economic sector (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87 0.85 (0.32) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87 0.85 (0.32) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Digital intensive industries 0.51 0.39 0.55 0.53 0.47 0.54 0.50 0.34 0.56
Liementary occupations 0.14 0.27 0.10 0.15 0.33 0.10 0.14 0.23 0.10 (0.35) (0.44) (0.3) (0.35) (0.47) (0.3) (0.34) (0.23) 0.10 Distribution of employment by economic sector 0.11 0.12 0.11 0.07 0.09 0.07 0.14 0.13 0.15 Manufactures 0.32 (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87 0.85 (0.32) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Digital intensive industries 0.51 0.39 0.55 0.53 0.47 0.54 0.50 0.34 0.56 (0.5) (0.49) (0.5) (0.5) (0.5) (0.5) (0.5) (0.48) (0.5) Skill measures 0.00 -0.57 0.20 -0.03 -0.70 0.14 0.03
Distribution of employment by economic sector (0.33) (0.44) (0.3) (0.33) (0.47) (0.3) (0.34) (0.29) Distribution of employment by economic sector (0.32) (0.31) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.32) Manufactures (0.32) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87 0.85 (0.32) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Digital intensive industries 0.51 0.39 0.55 0.53 0.47 0.54 0.50 0.34 0.56 (0.5) (0.5) (0.5) (0.5) (0.5) (0.5) (0.5) (0.48) (0.5) Skill measures ICT skills 0.00 -0.57 0.20 -0.03 -0.70 0.14 0.03 -0.49 0.24
Distribution of employment by economic sector 0.11 0.12 0.11 0.07 0.09 0.07 0.14 0.13 0.15 Manufactures (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87 0.85 Digital intensive industries 0.51 0.39 0.55 0.53 0.47 0.54 0.50 0.34 0.56 Kill measures (0.5) (0.49) (0.5) (0.5) (0.5) (0.5) (0.5) (0.48) (0.5)
Manufactures 0.11 0.12 0.11 0.07 0.05 0.07 0.14 0.13 0.13 (0.32) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.37 0.85 Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87 0.85 Digital intensive industries 0.51 0.39 0.55 0.53 0.47 0.54 0.50 0.34 (0.35) Skill measures (0.5) (0.49) (0.5) (0.5) (0.5) (0.5) (0.5) (0.48) (0.5)
Services 0.82 (0.32) (0.31) (0.26) (0.26) (0.26) (0.33) (0.34) (0.35) Services 0.89 0.88 0.89 0.93 0.91 0.93 0.86 0.87 0.85 (0.32) (0.32) (0.31) (0.26) (0.28) (0.26) (0.35) (0.34) (0.35) Digital intensive industries 0.51 0.39 0.55 0.53 0.47 0.54 0.50 0.34 0.56 (0.5) (0.5) (0.5) (0.5) (0.5) (0.5) (0.5) (0.5) 0.54 0.50 0.34 0.56 Skill measures 0.00 -0.57 0.20 -0.03 -0.70 0.14 0.03 -0.49 0.24
Services 0.69
Digital intensive industries 0.51 0.39 0.55 0.53 0.47 0.54 0.50 0.34 0.56 Skill measures 0.00 -0.57 0.20 -0.03 -0.70 0.14 0.03 -0.49 0.24
Digital intensive industries 0.51 0.53 0.55 0.55 0.54 0.55 0.55 (0.5)
Skill measures 0.00 -0.57 0.20 -0.03 -0.70 0.14 0.03 -0.49 0.24
CT skills 0.00 -0.57 0.20 -0.03 -0.70 0.14 0.03 -0.49 0.24
Readiness to learn 0.00 - 0.32 0.11 0.00 - 0.34 0.10 - 0.01 - 0.31 0.11
Management and communication 0.00 - 0.48 0.17 - 0.01 - 0.58 0.14 0.01 - 0.42 0.19
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Self-organization 0.00 -0.16 0.06 -0.03 -0.18 0.01 0.07 -0.15 0.09
Marketing and accounting $0.00 - 0.26 = 0.09 = 0.08 - 0.22 = 0.15 - 0.06 - 0.28 = 0.03$
(1) (1) (0.99) (1) (1.04) (0.98) (1) (0.97) (0.99)
STEM-guantitative skills 0.00 -0.38 0.13 -0.15 -0.62 -0.02 0.12 -0.24 0.27
(1) (0.76) (1.04) (0.92) (0.6) (0.95) (1.04) (0.8) (1.09)
Observations 3,242 820 2,422 1,434 298 1,136 1,808 522 1,286

Notes: Table reports descriptive statistics on the estimation sample extracted from the 2014 PIAAC (Chile) and 2012 (Bolivia, Colombia) and 2013 (El Salvador) STEP survey. Standard deviation is in parentheses. Low-educated: Lower secondary education or lower. High-educated: Upper secondary education or above. ICT: Information and communication technologies.

Figure 1: Distribution of Labor Market States



Women

Source: Own elaboration based on PIAAC and STEP surveys.



Figure 2: Skills in the Digital Economy

Source: Own elaboration based on PIAAC and STEP surveys.

Figure 3: Returns to Skills



Source: Own elaboration based on PIAAC and STEP surveys.

Note: The figure reports the coefficient estimate of those individual skill that are relevant to the digital economy in the log hourly wage regression defined in equation (1). It represents the percentage change in hourly wages of one standard deviation increase in the skill measure. The dot and the diamond are the point estimates, while the vertical lines show the 90% confidence interval of the estimate. The complete specification is defined in Section 4, equation (1). Complete results are available in Table B1 in Appendix B.



Figure 4: Returns to Skills in Sectors with Varying Digital Intensity: Extra Returns for Working in Digital-Intensive Sectors

Source: Own elaboration based on PIAAC and STEP surveys.

Note: The figure reports the estimate of the coefficient capturing the interaction between the individual skill relevant to the digital economy and a dummy equal to 1 if the sector is a digital-intensive sector. The estimate is obtained in the log hourly wage regression defined in equation (1). It represents the percentage change in hourly wages of a one standard deviation increase in the skill measure of a digital-intensive sector with respect to a nondigital-intensive sector. The dot and the diamond are the point estimates, while the vertical lines show the 90% confidence interval of the estimate. The complete specification is defined in Section 4, equation (1). Complete results are available in Table B2 in Appendix B.



Figure 5: Heterogeneous Effects by Education: Extra Returns to Skills for High-Educated Workers

Source: Own elaboration based on PIAAC and STEP surveys.

Note: The figure reports the coefficient estimate for the interaction between the individual skill relevant to the digital economy and a dummy equal to 1 if the individual has completed upper secondary education or above. The estimate is obtained in the log hourly wage regression defined in equation (1). It represents the percentage change in hourly wages of a one standard deviation increase in the skill measure for individuals with at least upper secondary education completed with respect to individuals who have not completed that level. The dot and the diamond are the point estimates, while the vertical lines show the 90% confidence interval of the estimate. The complete specification is defined in Section 4, equation (1). Complete results are available in Table B3 in Appendix B.

Figure 6: Returns to Skills: Correcting for Selection in the Labor Market



Source: Own elaboration based on PIAAC and STEP surveys.

Note: The figure reports the same estimates as Figure 3 but corrects for selection in the labor market for women. See Section 4.3 for additional details. We correct for selection only on women because men have a sufficiently high participation rate. Complete results are available in Table B4 in Appendix B.

Figure 7: Heterogeneous Returns to Skills: Correcting for Selection in the Labor Market



Extra Returns for Working in Digital-Intensive Sectors

Extra Returns for High-Educated Workers



Source: Own elaboration based on PIAAC and STEP surveys.

Note: The figure reports the same estimates as Figure 4 (top panel) and Figure 5 (bottom panel) but corrects for selection in the labor market for women. See Section 4.3 for additional details. We correct for selection only on women because men have a sufficiently high participation rate. Complete results are available in Tables B5 and B6 in Appendix B.



Figure 8: Contribution of the Skills' Returns and Endowments to the Gender Wage Gap

Source: Own elaboration based on PIAAC and STEP surveys.

Note: The figure reports results based on the Oaxaca-Blinder decomposition defined in equation (3). The *Endowments* correspond to the first term on the right-hand side of equation (3) and represent the contribution to the overall gap due to gender differences in endowments (quantities) of each individual skill relevant to the digital economy. The *Coefficients* correspond to the second term on the right-hand side of equation (3) and represent the contribution to the overall gap due to gender differences in returns (prices) to those skills. The dot and the diamond are the point estimates, while the vertical lines show the 90% confidence interval of the estimate. Complete results are available in Table B10 in Appendix B.

APPENDIX A: Data Appendix

Table A1. Skill-Related Questions from the PIAAC and STEP Surveys

Skill measure				:
	Questions used	Metric	Questions used	Metric
	Frequency of spreadsheets use, e.g. Excel	1 (never) to 5 (every day)	Regular use of telephone, mobile phone, pager, or other communication device	Yes/No
	Frequency of programming language use	1 (never) to 5 (every day)	Regular use of barcode reader	Yes/No
	Frequency of transactions through Internet (banking, selling/buying)	1 (never) to 5 (every day)	Use of a computer	Yes/No
	Frequency of email use	1 (never) to 5 (every day)	Frequency of computer use	1 (every day) to 4 (almost never); reverse order
	Frequency of simple Internet use	1 (never) to 5 (every day)	Use of email	Yes/No
	Frequency of text-processing program use, e.g. Word	1 (never) to 5 (every day)	Search info on the Internet	Yes/No
	Frequency of real-time discussions through ICT devices	1 (never) to 5 (every day)	Data entry	Yes/No
ICT skills	Frequency of reading letters, emails, memos	1 (never) to 5 (every day)	Word processing	Yes/No
	Frequency of writing letters, emails, memos	1 (never) to 5 (every day)	Use of spreadsheets	Yes/No
	Level of computer use required for the job	1 (normal) to 3 (complex)	Use of databases	Yes/No
			Use of software packs, design of websites, or programming	Yes/No
			Use of macros and complex equations	Yes/No
			Use of accounting or financial software	Yes/No
			Use of presentation or graphics software	Yes/No
			Use of website designing programs	Yes/No
			Use of CAD software	Yes/No
			Use of of statistical or other type of analysis	Yes/No
			Use of software programming	Yes/No
			Use of computer network management	Yes/No
	Frequency of negotiating with people inside or outside the firm or organization	1 (never) to 5 (every day)	Contacting people other than co-workers (e.g., customers, clients, students, members of the	Yes/No
			public) Meet or interact for at least 10-15 minutes at a time	1 (few times) to 10 (most of
Managerial and	Frequency of planning activities of others	1 (never) to 5 (every day)	with a customer, client, student or the public	the time)
communication	- - - - -		Make formal presentations to clients or colleagues	
	Frequency or instructing and teaching people	Thever) to 5 (every day)	to provide information of persuade them of your point of view	YES/NO
	Frequency of advising people	1 (never) to 5 (every day)	Direct and check the work of other workers	Yes/No
	Frequency of persuading or influencing others	1 (never) to 5 (every day)	(achor kiac)	

Table A1 cont.

1 III 0100 I				
Chill mooon inco	PIAAC		STEP	
	Questions used	Metric	Questions used	Metric
	I like to get to the bottom of difficult things	1 (not at all) to 5 (to a very high extent)	Engage in tasks that involve 30 or more minutes of thinking	1 (never) to 5 (every day)
	If I don't understand something, I look for additional information to make it clearer	1 (not at all) to 5 (to a very high extent)	Learning new things	1 (every day) to 5 (rarely or never); reverse order
	When I come across something new, I try to relate to what I already know	1 (not at all) to 5 (to a very high extent)	Come up with ideas other people haven't thought of before	 almost always) to 4 (almost never); reverse order
mod of providend	When I hear or read about new ideas, I try to relate them to real life situations to which they might apply	1 (not at all) to 5 (to a very high extent)	Finish whatever you begin	 (almost always) to 4 (almost never); reverse order
and creative problem- solving	I like learning new things	1 (not at all) to 5 (to a very high extent)	Think carefully before you make an important decision	1 (almost always) to 4 (almost never); reverse order
	I like to figure out how different ideas fit together	1 (not at all) to 5 (to a very high extent)	Work very hard, such as continuing to work when others stop to take a break	 (almost always) to 4 (almost never); reverse order
			Be very interested in learning new things	1 (almost always) to 4 (almost never); reverse
			Ask for help when don't understand something	order 1 (almost always) to 4 (almost never); reverse order
	Extent of own planning of the task sequences	1 (not at all) to 5 (to a very high extent)	Freedom to decide how to do your work in your own way, rather than following a fixed procedure or a supervisor's instructions	1 (no freedom) to 10 (complete freedom)
Self-organization	Extent of own planning of style of work	1 (not at all) to 5 (to a very high extent)		
	Extent of own planning of speed of work	1 (not at all) to 5 (to a very high extent)		
	Extent of own planning of working hours	1 (not at all) to 5 (to a very high extent)		
	Frequency of reading financial invoices, bills, etc.	1 (never) to 5 (every day)	Read bills or financial statements	Yes/No
Marketing and	Frequency of calculating prices, costs, or budgets	1 (never) to 5 (every day)	Calculate prices or costs	Yes/No
accounting	Frequency of using a calculator Frequency of selling a product or a service	1 (never) to 5 (every day) 1 (never) to 5 (every day)		
	Frequency of preparing charts and tables	1 (never) to 5 (every day)	Measure sizes, weights, distances	Yes/No
STEM quantitative	Frequency of use of simple algebra and formulas	1 (never) to 5 (every day)	Use or calculate fractions or decimals	Yes/No
skills	Frequency of use of complex algebra and statistics	1 (never) to 5 (every day)	Perform any other multiplication or division	Yes/No
Source. Our	elaboration based on DIAAC and S	TFD surveys		

source: Uwn elaboration based on PIAAC and STEP surveys. Notes: *Reverse order* means we reversed the order of the categories such that a higher value mean higher frequency of the task.

Table A2. Average Num	per of Operation	Robots	Over 2011-	2014
-----------------------	------------------	--------	------------	------

	Latin	USA	Western
	America	00/1	Europe
Agriculture, forestry and fishing	2.6	54.0	30.9
Mining and quarrying	1.9	11.0	8.1
Manufacturing	1,457	146,608	22,023
Electricity, gas, water supply	0.3	7.8	12.7
Construction	1.5	105.3	60.9
Education and Research development	5.4	328.3	212.5
All other non-manufacturing branches	3.5	189.3	26.4

Source: Own elaboration based on International Federation of Robotics.

Note: For Latin America and Western Europe, the table shows the cross-country average from 2011 to 2014.

APPENDIX B: Complete Estimation Results

Table B1. Returns to Skills

				Depende	ent variable: I	-og of hourly	wages (in Ut	SD at PPP of	2005)			
	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men
Age	-0.00736	0.0284	0.00126	0.0393	-0.00166	0.0346	0.000304	0.0413	0.0044	0.0417	-0.00123	0.0362
	[0.0105]	[0.00799]***	[0.0107]	0.00812]***	[0.0106]	[0.00798]***	[0.0107]	0.00810]***	[0.0108]	[0.00809]***	[0.0107] [***[66700.0
Age sq	0.00015	-0.000265	3.69E-05	-0.00039	7.30E-05	-0.000344	3.62E-05	0.000423	-7 98E-06	-0.000425	7.14E-05	-0.000355
	[0.000137]	0.000101]**	[0.000141]	0.000103]**	[0.000138]	0.000101]**	[0.000140] [0.000103]**	[0.000141]	0.000103]**	[0.000141] (0.000101]**1
=1 if high-educated	0.293	0.254	0.398	0.349	0.322	0.291	0.405	0.357	0.424	0.361	0.377	0.308
	[0.0435]***	[0.0348]***	[0.0415]***	[0.0340]***	[0.0432]***	[0.0341]***	[0.0407]***	[0.0336]***	[0.0413]***	[0.0340]***	[0.0426]***	[0.0339]***
	0.0197]***	0.1/4 [0.0156]***										
Readiness to learn			0.066 r0.01e01***	0.0725 0.01461***								
			0.0103	0+10.0								
Management and communication					0 142 [0 0190]***	0.141 [0.0150]***						
Self-organization							0.0941	0.0798				
							[0.0175]***	[0.0153]***				
Marketing and accounting									-0.000929	0.0549		
									[0.0184]	[0.0155]***		
STEM-quantitative											0.0852	0.152
Constant	0.536	0.0145	0.298	-0.243	0.407	-0.11	0.364	-0.119	0.23	-0.283	0.369	0.164
	[0.191]***	[0.152]	[0.194]	[0.153]	[0.192]**	[0.152]	[0.197]*	[0.151]	[0.193]	[0.154]*	[0.193]*	[0.152]
Observations	1,434	1,808	1,434	1,808	1,434	1,808	1,427	1,798	1,434	1,808	1,434	1,808
R-squared	0.335	0.353	0.303	0.318	0.325	0.341	0.309	0.316	0.297	0.314	0.306	0.349
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Source: Own elaboration	on based	on PIA	AC and	STEP s	urveys.							

Notes: Ordinary least squares regressions. Robust standard errors in brackets. *** p<0.01, ** p<0.05, * p<0.1 PPP: Purchasing power parity.

I able DZ. Returns	TINC 01	ac III S	CIUIS V	V IUI V	art variable ¹	DIGITAI		SILY SD at PPP o	f 2005)			
	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men
Age	-0.00755	0.0276	0.00156	0.0394	-0.0014	0.0343	0.000172	0.0415	0.00419	0.0418	-0.00114	0.0362
	[0.0105]	[0.00801]***	[0.0107]	[0.00812]***	[0.0106]	0.00798]***	[0.0107]	0.00810]***	[0.0108]	0.00810]***	[0.0107]	.00799]***
Age sq	0.000152	-0.000255 In non1011**	3.42E-05	0.00039	6.96E-05	0.00034	4.01E-05	-0.000426	-3.16E-06 In 0001411 I	0.000425	7.10E-05	0.000355
=1 if high-educated	0.293	0.263	0.397	0.35	0.321	0.299	0.396	0.357	0.421	0.363	0.376	0.305
	[0.0436]***	[0.0352]***	[0.0417]***	[0.0344]***	[0.0433]***	[0.0342]***	[0.0412]***	[0.0342]***	[0.0413]***	[0.0344]***	[0.0427]***	0.0344]***
=1 if digital intensive sector	-0.0123 [0 0356]	-0.0436 [0.0309]	0.0224 [0.0357]	-0.000451 [0 0315]	0.00673 [0 0353]	-0.0363 [0 0314]	0.0204 [0.0357]	0.0036 [0.0314]	0.0236 [0 0371]	-0.0032 [0 0319]	0.0204 [0 0362]	0.0108 [0.0308]
ICT skills	0.173	0.161										
	[0.0301]***	[0.0219]***										
ICI skills =1 if diaital-intensive sector	10 03811	0.0341 [0.0302]										
Readiness to learn			0.0579	0.0686								
			""[0c200]	0.0198]								
			0.0103	0.00803								
=1 IT digital-intensive sector			0.0362	0.0284	0.450	007.0						
Management and communication					0.133 In 02571***	U.139 [0 0213]***						
Management and communication *					-0.0218	0.0105						
=1 if digital-intensive sector					0.0356]	0.03001						
Self-organization							0.13	0.0604				
:							[0.0246]***	[0.0218]***				
Self-organization							-0.0689	0.0389				
 If digital-intensive sector Markating and accounting 							[0.0352]*	[0.0305]				
									[0.0275]*	[0.0230]*		
Marketing and accounting									0.0794	0.0276		
=1 if digital-intensive sector									[0.0370]**	[0.0311]		
STEM-quantitative											0.07 In 03171**	0.15
STEM-quantitative											0.0268	0.00342
=1 if digital-intensive sector											[0.0392]	[0.0298]
Constant	0.546	0.0447	0.282	-0.245	0.402 ro 4021**	-0.0874	0.365 0.4001*	-0.121	0.212	-0.285	0.356 ro 1041*	-0.169 ro.1521
	0.130]	[0.134]	0.134]	0.134	0.130	[ec1.0]	0.199]	[zei .0]	[0.130]	[cc1 .0]	0.134]	[cc1.0]
Observations	1,434	1,808	1,434	1,808	1,434	1,808	1,427	1,798	1,434	1,808	1,434	1,808
R-squared	0.335	0.354	0.303	0.318	0.325	0.342	0.311	0.317	0.3	0.315	0.306	0.349
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Source: Our elaborati	pased no	I on DI A	AC and	CTED o	STIDITALL							

Table B2. Returns to Skills in Sectors with Varying Digital Intensity

I aute DJ. Netullis I	IIIYC OI	s. nele	IUBCII	COUS E	T CCIP I	N EUUN	<u>catlull</u>					
	,			Depende	int variable: L	og of hourly	vages (in US	D at PPP of	2005)			
	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men
Age	-0.00781	0.0272 ro.007071***	0.0014	0.0396	-0.00208	0.035	0.000113	0.0421	0.00436	0.0418	-0.00167	0.0361
Age sq	0.000155	0.00025	3.47E-05	-0.000395	7.81E-05	-0.00035	3.92E-05	0.000434	7.45E-06	-0.000427	7.67E-05	-0.000354
	[0.000137]	[0.000101]**	[0.000141]	0.000104]**	0.000138] (D.000100]**1	[0.000140] [0	000102]**	[0.000142] (0.000102]**	[0.000141] C	.000101]**
=1 if high-educated	0.336	0.309	0.394	0.354	0.338	0.341	0.411	0.364	0.424	0.379	0.407	0.319
ICT ski l s	0.104	0.0491	0.0435)***	[0.0345]***	[0.0491]***	0.0347]***	0.0420]***	0.0337]***	[0.0415]***	0.0340]***	0.0492]***	0.0339]***
	[0.0496]**	[0.0311]										
ICI SKIIS -1 if high oct ingtod	U.U/3/	901.0 ***I8360.01										
Readiness to learn	[ot oo.o]	00000	0.0796	0.0498								
			[0:0330]**	[0.0222]**								
Readiness to learn * -1 if high ordination			-0.0178	0.035								
- I II IIII reducated Management and communication			[/eco.o]	[0020.0]	0.114	-0.00144						
					[0.0388]***	[0.0273]						
Management and communication * =1 if high-educated					0.035 [0.0442]	0.19 [0.0327]***						
Self-organization							0.0591	0.00947				
Self-organization							0.0454	0.107				
=1 if high-educated							[0.0402]	0.0310]***				
Marketing and accounting									-0.00326 10.03171	-0.0149 0.02611		
Marketing and accounting									0.00302	0.0964		
=1 if high-educated									[0.0380]	[0.0318]***	0.0377	8780.0
o i Eivi-quai ilitative											[0.0490]	0.0316]***
STEM-quantitative =1 if hinh-educated											0.0522 [0.0533]	0.0813 In 03591**
Constant	0.504	-0.0242	0.301	-0.253	0.398	-0.18	0.361	-0.137	0.23	-0.303	0.347	0.176
	[0.192]***	[0.152]	[0.194]	[0.154]	[0.193]**	[0.150]	[0.197]*	[0.151]	[0.194]	[0.152]**	[0.192]*	[0.152]
Observations	1,434	1,808	1,434	1,808	1,434	1,808	1,427	1,798	1,434	1,808	1,434	1,808
R-squared	0.336	0.359	0.303	0.319	0.325	0.352	0.31	0.321	0.297	0.317	0.306	0.35
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table B3. Returns to Skills: Heterogeneous Effects by Education

Countries FE Ves Ves I res Countries FE Ves Ves I res Countries FE Verses: Own elaboration based on PIAAC and STEP surveys. Source: Own elaboration based on PIAAC and STEP surveys. Notes: Ordinary least squares regressions. Robust standard errors in brackets. *** p<0.01, ** p<0.05, * p<0.1 PPP: Purchasing power parity.

I auto DT. Notullis tu	ONITIO.	CUILC	vuiig i				י וטטאי	MAINU	د			
	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men
Regression equation												
Dependent variable: Log of hourly wage:	s (in USD at P	PP of 2005)										
Age	0.00349	0.0284	0.00845	0.0393	0.00518	0.0346	0.00772	0.0413	0.0131	0.041/	0.00/48	0.0362
	[0.0163]	0.00799]***	[0.0148]	[0.00812]***	[0.0143]	0.00798]***	[0.0139]	0.00810]***	[0.0153]	0.00809]***	0.0151]	
Age sq	9.47E-05	-0.000265	2.15E-05	-0.00039	6.28E-05	-0.000344	1.70E-05	0.000423	-3.44E-05	-0.000425	4.02E-05	0.000355
	[0 000208] (0.000101]**	[0.000189]	0.000103]**	[0.000182] (0.000101]**	[0.000177] (0.000103]**	[0.000195] (0.000103]**	0.000193] 0	.000101]**
=1 if high-educated	0.298 rn n7ne1***	0.254 In 03481***	0.419 ID DE1E1***	0.349 ro o2401***	0.309 ro ne101***	0.291 ro 02411***	0.432 ro.05741***	0.357 In nagel***	0.437 In ne371***	0.361 ro o2401***	0.393 0.06411***	0.308
ICT skills	0.188 0.188 0.02891***	[0.0340] 0.174 [0.0156]***	[0] 00 0]	[0+c0.0]	[e100.0]	[1+00.0]	[+ / co. o]	[0000.0]	[/000.0]			6000.0
Readiness to learn			0.0697 0.0561***	0.0725 0.01461***								
Management and communication					0.176	0.141 0.04501***						
Self-organization					[nczn]	[nc1n]	0.0754 0.0231***	0.0798 0.01631***				
Marketing and accounting							[0770]	[00100]	0.0271	0.0549		
STEM-quantitative									[0.0256]	[0.0155]***	0.0948	0.152
Constant	0.261 [0.306]	0.0145 [0.152]	0.0835 [0.277]	-0.243 [0.153]	0.219 [0.268]	-0.11 [0.152]	0.117 [0.261]	-0.119 [0.151]	-0.0187 [0.285]	-0.283 [0.154]*	0 0276]*** 0 113 [0 283]	0.0153]*** -0.164 [0.152]
Observations	1,220	1,808	1,220	1,808	1,220	1,808	1,213	1,798	1,220	1,808	1,220	1,808
Selection equation Dependent variable: =1 if actively partici Aoe	ipating in labor 0.188	market	0.188		0.188		0.188		0.188		0.188	
Age so	[0.0522]*** -0.00185		[0.0522]*** -0.00185		[0.0522]*** -0.00185		[0.0522]*** -0.00185		[0.0522]*** -0.00185		0.0522]*** -0.00185	
-	0.000645]***		0.000645]***		0.000645]***		0.000645]***	- 2	0.000645]***		.000645]***	
=1 if high-educated	-0.117 11/2011		-0.117		-0.117		-0.117		-0.117		-0.117 0.2211	
Number of children	0.21		0.21		0.21		0.21		0.21		0.21	
	[0.0903]**		[0.0903]**		[0.0903]**		[0.0903]**		[0.0903]**		[0.0903]**	
Lambda	0.95		0.86		0.84		0.81		06.0		0.88	
std. error.	0.40		0.36		0.35		0.34		0.37		0.37	
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
		1 I L										

Table B4. Returns to Skills: Correcting for Selection in the Labor Market

Countries FE verse verse verse verse verses. Source: Own elaboration based on PIAAC and STEP surveys. Notes: Heckman selection models. Robust standard errors in brackets. *** p<0.01, ** p<0.05, * p<0.1 PPP: Purchasing power parity.

Digital Intensity		,	,								•)
	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men
Regression equation Dependent variable: Log of hourly wages (in USD at PPF Age	P of 2005) 0.0033 10.0033	0.0276	0.00861	0.0394	0.00543	0.0343	0.00737	0.0415	0.013	0.0418	0.00753	0.0362
Age sq	9.66E-05	0.000255	2.02E-05	0.00039	5.97E-05	0.00/98] -0.00034	[0.0132] [2.30E-05 0.0004601 [0.000426	3.17E-05	0.000425	4 04E 05	-0.000355
=1 if high-educated	0.299	0.263	0.417	0.35	0.308	0.299	0.423	0.357	0 433	0.363	0.391	0.305
=1 if digitaLintensive sector	0.024 0.024 0.0550	0.0436 0.0436	0.0287	0.000451 0.000451	0.00665	0.0363 0.0363 0.0341	0.0236 0.0236	0.0036 0.0036 0.03141	0.0135 0.0135 0.05101	0.0032	0.0227 0.0227	0.0108 0.0108 0.03081
ICT skills	0.182	0.161 0.161	[00+0-0]	[0100-0]	[- <u></u>	F	[(p+p-p)	E 00.0	6 00 0	[c] 00-0]	[000000]	[0000-0]
ICT skills =1 if digital-intensive sector Readiness to learm	[0.0555]	0.0341 [0.0302]	0.0668	0.0686								
Readiness to learn * =1 if digitaLintensive sector Management and communication			[0.00576 [0.0495]	0.00803 0.00803 [0.0284]	0.183	0.139						
Management and communication * = 1 if digital-intensive sector Self-organization					[0.0473] 0.014 [0.0473]	[0.02.03] 0.0105 [0.0300]	0.11	0.0604				
Self-organization * =1 if digital-Intensive sector Marketing and accounting							[0.0301] *** [0.0688	[0.0218] 0.0389 [0.0305]	-0.0302	0.0409		
Marketing and accounting * =1 if digital-intensive sector STEM-quantitative									[0.0515]**	[0.0230] 0.0276 [0.0311]	0.0758	0.15
STEM-quantitative * =1 fi digital-intensive sector Constant	0.274 [0.312]	0.0447 [0.154]	0.066 [0.275]	-0.245 [0.154]	0.213 [0.268]	-0.0874 [0.153]	0.121 [0.250]	-0.121 [0.152]	-0.0377 [0.285]	-0.285 [0.155]*	[0.0400]* 0.0335 [0.0526] 0.0989 [0.282]	[0.0227] 0.00342 [0.0298] -0.169 [0.153]
Observations	1,220	1,808	1,220	1,808	1,220	1,808	1,213	1,798	1,220	1,808	1,220	1,808
Selection equation Dependent variable: =1 if actively participating in labor m Age	arket 0.188		0.188		0.188		0.188		0.188		0.188	
– s. ∀đe sd	0.00185		0.00185		0.00185		0.00185	£	0.00185		0.00185	
=1 if high-educated	0.117 -0.117 10.2211	<u> </u>	-0.117 -0.117 10.2211	±	-0.117 0.221		-0.117 0.117	<u>.</u>	-0.117 -0.231	<u> </u>	-0.117 0.117	
Number of children	-0.21 -0.21 [0.0903]**		-0.21 -0.21 [0.0903]**		0.21 -0.21 [0.0903]**		0.21 0.21 [0.0903]**		0.21 -0.21 [0.0903]**		0.21 -0.21 [0.0903]**	
Observations	1,220		1,220		1,220		1,220		1,220		1,220	
Lambda std. error.	0.96 0.40		0.86 0.36		0.83 0.35		0.77 0.32		0.89 0.37		0.88 0.36	
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table B5. Returns to Skills: Correcting for Selection in the Labor Market. Sectors with Varying . T T

Countries FE very countries FE very countries FE very solution based on PIAAC and STEP surveys. Source: Own elaboration based on PIAAC and STEP surveys. Notes: Heckman selection models. Robust standard errors in brackets. *** p<0.01, ** p<0.05, * p<0.1 PPP: Purchasing power parity.

by Education												
	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men
Regression equation Dependent variable: Log of hourly wages	s (in USD at F	PP of 2005	-									
Age	0.00332 I0.01631	0.0272	0.00839 [0.0148]	0.0396 0.08141***	0.00435 [0.0143]	0.035	0.00768 I0.01401 I	0.0421 0.008041***	0.0126 0.01521 F	0.0418 0.008051***	0.00686 I0.01521	0.0361
Age sq	9.63E-05	0.00025	2.24E-05	0.000395	7.30E-05	0.00035	1.83E-05	-0.000434	-2.84E-05	-0.000427	4.75E-05	0.000354
=1 if high-educated	0.332	0.309	0.422	0.354	0.343	0.341 0.341	0.436	0.364	0.441 0.6381***	0.379	0.425	0.319
ICT skills	0.136 0.136 0.07571*	0.0491 0.0491 0.03111	0.0002	[oto:o]	[+ /oo.o]	[top o]	[2000.0]	[,,,,,,,,]	[0000.0]	[otco.o]	[0:00:0]	[acco.o]
ICT skills =1 if high-educated	0.0609	0.159 0.159 0.03581***										
Management and communication			0.063 0.05021	0.0498 0.0221**								
Management and Communication * =1 if high-educated Self-organization			[0.0584]	[0.0288]	0.116	-0.00144						
Self-Organization =1 if high-educated Marketing and accounting					[/ 560.0] 0.0766 [7090.0]	[0.0273] 0.19 [0.0327]***	0.0529	0.00947				
Self-Organization =1 if high-educated							[0.0451] 0.03 [0.0519]	[0.0243] 0.107 [0.0310]***	C6100 0	0110		
									[0.0499]	-0.0143 [0.0261]		
Marketing and Accounting =1 if high-educated STEM-quantitative									0.0345 [0.0577]	0.0318]***	0.0426	0.0848 0.03461***
STEM-quantitative =1 if high-educated Constant	0.226 [0.310]	-0.0242 [0.152]	0.0818 [0.278]	-0.253 [0.154]	0.195 [0.268]	-0.18 [0.150]	0.113 [0.263]	-0.137 [0.151]	-0.0175 [0.284]	-0.303 [0.152]**	(0.097.9] 0.0583 0.0919] 0.0944 [0.285]	0.0813 0.0813 [0.0359]** -0.176 [0.152]
Observations	1,220	1,808	1,220	1,808	1,220	1,808	1,213	1,798	1,220	1,808	1,220	1,808
Selection equation Dependent variable: =1 if actively partici Age	oating in labol 0.188	r market	0.188		0.188		0.188		0.188		0.188	
Age sq	[0.0522]*** -0.00185	2	[0.0522]*** -0.00185	e	[0.0522]*** -0.00185	2	[0.0522]*** -0.00185	2	[0.0522]*** -0.00185	2	[0.0522]*** -0.00185	
=1 if high-educated	-0.117	<u> </u>	-0.117		-0.117 -0.117	<u> </u>	-0.117		-0.117	<u> </u>	-0.117 -0.117	
Number of children	-0.21 -0.21 [0.0903]**		-0.21 -0.21 [0.0903]**		-0.21 -0.21 [0.0903]**		-0.21 -0.21 [0.0903]**		-0.21 -0.21 [0.0903]**		-0.21 -0.21 [0.0903]**	
	U OF		0.87		0 83		0 8.0		08 0		88 0	
std. error.	0.40		0.36		0.35		0.34		0.37		0.37	
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table B6. Returns to Skills: Correcting for Selection in the Labor Market. Heterogeneous Effects

Countries FE Very Countries FE Very Countries FE Very Countries FE Very Source: Own elaboration based on PIAAC and STEP surveys. Notes: Heckman selection models. Robust standard errors in brackets. *** p<0.01, ** p<0.05, * p<0.1 PPP: Purchasing power parity.

	Depend	ent variable:	Log of hourl	y wages (in l	JSD at PPP	of 2005)
=1 if female	-0.174	-0.194	-0.185	-0.191	-0.198	-0.159
	[0.0226]***	[0.0232]***	[0.0228]***	[0.0233]***	[0.0234]***	[0.0232]***
Age	0.013	0.0228	0.019	0.0237	0.0255	0.0201
	[0.00637]**	[0.00649]***	`[0.00638]** [*]	[0.00646]***	[0.00649]***	[0.00640]***
Age sq	-8.64E-05	-0.000205	-0.000165	-0.000226	-0.000243	-0.000171
	[8.17e-05]	[8.37e-05]**	*[8.18e-05]**	[8.30e-05]**	[8.34e-05]**	'[8.24e-05]**
=1 if high-educated	0.267	0.369	0.303	0.376	0.385	0.335
	[0.0270]***	[0.0262]***	[0.0266]***	[0.0259]***	[0.0262]***	[0.0264]***
ICT skills	0.176					
	[0.0154]***					
ICT skills * =1 if female	-0.00628					
	[0.0236]					
Readiness to learn		0.0704				
		[0.0144]***				
Readiness to learn * =1 if female		-0.00289				
		[0.0229]				
Management and communication		[]	0 143			
			[0 0150]***			
Management and communication * =1 if female			0.000503			
			[0 0230]			
Self-organization			[0.0200]	0 0784		
				[0 0153]***		
Self-organization * =1 if female				0.0133		
				[0 0233]		
Marketing and accounting				[0.0200]	0.0517	
Marketing and accounting					10 01541***	
Marketing and accounting * =1 if female					_0 0/33	
					10 02351*	
STEM quantitative					[0.0200]	0 140
STEM-quantitative						0.149
STEM quantitative * -1 if female						0.0503
						-0.0393
Constant	0 325	0.0861	0 201	0.16	0.0348	0 142
Constant	0.325	0.0001	0.201	0.10	0.0340	0.142
	[0.119]	[0.120]	[0.119]	[0.110]	[0.120]	[0.119]
Observations	2 040	2 240	2 0 4 0	2 225	2 0 4 0	2 242
Diservations Diservations	3,242 0.245	२,८ 4 ८ ०२१	3,∠ 4 ∠ 0.222	3,223 0,212	3,242 0.205	२,८4८ ०.२२०
n-syualeu	0.340	0.31	0.333	0.312	0.305	0.320

Table B7. Returns to Skills: Robustness Analysis. Constraining All the Coefficients to Be the Same for Men and Women with the Exception of Their Returns to Skills

Source: Own elaboration based on PIAAC and STEP surveys. Notes: Ordinary least squares regressions. Robust standard errors in brackets. *** p<0.01, ** p<0.05, * p<0.1 PPP: Purchasing power parity.

I aule Do. Neullis	LU JAIL	S. NUUI	asultes	S AIIal	VSIS. U	סוומות	o ginni	וו סכנו	UIS alle	I Occu	Dauous	
				Depende	ent variable: L	og of hourly	wages (in U	SD at PPP o	f 2005)			
	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men
Age	-0.0228	0.0159	-0.0205	0.0186	-0.0219	0.0167	-0.0226	0.0188	-0.0198	0.0193	-0.0207	0.0187
	[0.0101]**	[0.00741]**	[0.0102]**	[0.00739]**	[0.0102]**	[0.00733]**	[0.0102]**	[0.00735]**	[0.0102]*	0.00737]***	[0.0102]**	0.00736]**
Age sq	0.000319	-0.000122	0.000287	-0.000154	0.000305	-0.000133	0.000307	-0.000159	0.000273	-0.000164	0.000289	-0.000156
	[0.000131]**	· [9.32e-05] [0.000133]**	[9.30e-05]*	0.000132]**	[9.21e-05] [0.000132]**	[9.22e-05]* [0.000131]**	[9.25e-05]* [0.000131]**	9.26e-05]*
=1 if high-educated	0.181	0.176	0.199	0.195	0.186	0.183	0.192	0.191	0.204	0.199	0.197	0.185
	[0.0459]***	[0.0339]***	[0.0458]***	[0.0337]***	[0.0462]***	[0.0335]***	[0.0458]***	[0.0336]***	[0.0457]***	[0.0335]***	[0.0462]***	0.0334]***
ICT ski l s	0.0697 [0.0224]***	0.0713 [0.0167]***										
Readiness to learn		1	0.0154 [0.0179]	0.0235 [0.0133]*								
Management and communication					0.0516 [0.0198]***	0.0644 [0.0150]***						
Self-organization							0.0609	0.0402				
							[0.0167]***	0.0140				
Marketing and accounting									-0.0373 I0 01881**	0.0226 [0.0147]		
STEM-quantitative									[0.0143	0.0607
	900 1	, r	101	7077	000	1	000	777777777777777777777777777777777777777	1	(,	[0.0185] 1 722	0.0149]***
CONSIGNI	1.720 [0.258]***	1 143 [0 191]***	[0.259]***	[0.190]***	[0.258]***	1. 14 / [0. 190]***	1.009 [0.266]***	1.141 [0.187]***	0. 262]***	1.1.12 [0.192]***	1.722 [0.259]***	1 092 [0 190]***
Observations	1,434	1,808	1,434	1,808	1,434	1,808	1,427	1,798	1,434	1,808	1,434	1,808
R-squared	0.445	0.485	0.44	0.481	0.443	0.485	0.445	0.48	0.442	0.48	0.44	0.485
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Occupation and sector FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Source: Own elaborati	ion based	l on PIA	AC and	STEP s	urveys.							

Table B8. Returns to Skills: Robustness Analysis. Conditioning on Sectors and Occunations

Notes: Ordinary least squares regressions. Robust standard errors in brackets. *** p<0.01, ** p<0.05, * p<0.1 PPP: Purchasing power parity.

Ising Different Exc	lusic	on-Re	estric	tion	s. ST	ΈP (Coun	tries	Only														
nation Method:	JLS He	sckman Her	ickman Hei	sckman	H SIO	leckman H	eckman H ₆	sokman (JLS Heck	man Heck	cman Heckn	nan OLS	Heckma	an Heckma	n Heckman	SIO	Heckman	Heckman	Heckman	SIO	Heckman	Heckman	Heckman
usion-restriction:	2 D	imber of Gran hildren Pre	idparents E	Both	zŰ	umber of Gra Shildren Pr	ndparents 'esence	Both	Chik	ber of Grandp dren Presu	arents Both ence	_	Number Childre	of Grandpar n Present	ents Both		Number of Children	Grandparents Presence	Both		Number of Children	Grandparents Presence	Both
ression equation endent variable: Log of hourly vages (in USD at PPF -0.0	, of 2005) 10390 -0	1.0163 -0.	0355 -0.	0246 9.	.68e-06	0.0266 -(0439 -0	0.0349 -0.0	30174 -0.0;	226 -0.0	361 -0.02	78 -0.0015	59 -0.024	2 -0.040	9 -0.0337	0.000804	-0.0236	-0.0426	-0.0334	-0.00202	-0.0251	-0.0467	-0.0359
[0] 50 0.01	0130] [0 0200 0.0	0.0351) [0. 000271 0.0. voo <i>x</i> e41 0.0.0	0390] [0. 00517 0.0	0322] [C 00378 0.0	0.0131] [. 000140 0.	0.0356] [(000401 0.1	0.0395] [C 0.0623 0.0 0.0623 0.0	10327] [0. 100508 0.0.	0131] [0.0. 00159 0.000 01701 10.000	349] [0.03 0345 0.000 0449 0.000	386] [0.03 0517 0.000 04061 10.000	412 0.0014 412 0.00014 4101 0.00014	1] [0.035 48 0.0003 701 10.0003	50 0.005 50 0.0005	3] [0.0325] 35 0.000477	0.000127	0.000363	[0.0395] 0.000605 fo.000605	[0.0326] 0.000488 0.000488	0.000174	[0.0355] 0.000381 0.0004661	[0.0395] 0.000659 0.000659	[0.0326] 0.000520 0.0004401
10.01 high-educated 0.	233 [UI 233 [UI 233 [UI	0.136 0.	0.116 0	0.128 [0	0.313	0.167	0.149 [U.	0.158 0.0	223 0.00	994 0.00	0.094 857 0.094 857 0.094	43 0.316 10.0001	171 [0.0004	0.152	0.161	0.335	0.169	0.150	0.159	0.302	0.160	0.137	0.148
lu.u Informal workers -0	243 (0. 243 (0. 2611 (0.	0.227 0.0	0.226 -0	0.226 L	0.295 [0.	0.264	0.263	0.263 -C	262 -0.1	189 -0.1	188 -0.18	10.0531 38 -0.317 38 -0.317	7 10.0068	08000 - 0	0.0307	202:0-	-0.280	-0.279	-0.278	-0.292	-0.262	-0.260	-0.260
ber of children	in lene	n-nol [0:0+0		[00]	n]lenen	.n=100]			+n:n] [/.8+	*0.0]	140.0] [01		i teorio	1.40.0	[1140.0]	[men:n]	lon#n n	[00+0.0]	[antonin]	[tenen:n]	0.0404]	0.0404	[0.040.0]
dmother/grandfather present																							
skils	142 (0.112 0.	0.112 0	112																			
[U.U] diness to learn	263] [0.(n:n][znzn	n.ol).0432 (0.0282	0 000000	0278															
agement and communication				2	1	y] [e/10:0	1 [6/1.07		147 0.1	59 0.1	58 0.15												
organization								1.01	747] [0:02	70:0][en	nzn:n][en	0.055	1 0.082	7 0.082	0.0827								
ceting and accounting												6 70 ml			[11]0.0]	-0.0205	0.0332	0.0329	0.0333				
M-quantitative																nezn:n		0.0162	[2010.0]	0.0570	0.0513	0.0517	0.0516
o [02	604 (0.967 1 0.901] [0.	1.464 1 .998] [0	181 [0	0.509 1.243]** [1.217 0.914] [1	1.664 1.010]* [0	1.431 0 831]* [0.2	574 1.1 242]** [0.8	37 1.4 195] [0.9	1.26 183 1.26	9 0.566 3] [0.246]	* 1.195	1.631	1.445 [0.831]*	0.489 [0.244]**	1.163 [0.912]	1.652 [1.010]	1.414 [0.831]*	0.547 0.547 [0.244]**	[0.911] [0.911]	[1.011]* 1.748 [1.011]*	[0.031]*
ervations	15	3,184 3,	3,184 3.	1,184	715	3,184	3,184	3,184	715 3,1	84 3,1	84 3,18	14 708	3,173	3,173	3,173	715	3,184	3,184	3,184	715	3,184	3,184	3,184
ection equation sendent variable: =1 if actively participating in labor m. s	arket	.0	1218 0.	234	_	0.229	0.218 0	1,234	0.2	29 0.2	18 0.23	4	0.226	0.218	0.234		0.229	0.218	0.234		0.229	0.218	0.234
bs	9 0	0125]*** [0.0 00292 -0.0	0114]*** [0.0 00277 -0.0	00297	<u>e</u> Y	0125]*** [0.	0114]*** [0.0	00297	[0.01: -0.00	25]*** [0.011 0292 -0.00	14]*** [0.012] 0277 -0.002	297	[0.0125] -0.0025	2 -0.002	[0.0127] 7 -0.00297	<u> </u>	[0.0125]***	0.00277	[0.0127]*** -0.00297		[0.0125]*** -0.00292	[0.0114]*** -0.00277	[0.0127]*** -0.00297
fi high-educated		00161]***[0.00	0146 - 10.00	0162]		0.200	001467-70.0	0162]	0.000	161]-10.000	146]**[0.00016 222 0.19	22	0.200	0.222	0.195	ı — .	0.200	0.222	0.195		0.200	0.000146][0	0.195
nber of children	- - - - - - - - - - - - - - - - - - -	nni [anon		1010	2, " 9	n] _[anon		10100	00- 00-	797 vouu	90:0- 90:0-	12	620.0- 620.0-		[0160.0]		7970.0- 7800.0-	[inen:n]	7730.0-		7970.07 78000.01	[Incn:n]	-0.0677
ndmother/grandfather present	<u>10</u>	0.0] 0	0.181 0 594]*** [0.0	0.143 0.143 0609]**	2	10]	0.181 (0.594]*** [0.).143).143)609]**	Znini	0.1 [0.056	81 0.14 81 0.14 94]*** [0.060	9]**	[9770:0]	0.181	0.143		[0770].0]	0.181 [0.0594]***	0.143 0.143 [0.0609]**		[9770:0]	0.181 [0.0594]***	[0.0609]**
upation and sector dummies in reg. equation	No	No	No	No	No	No	No	No	No N	2 0	o No	No	No	No	No	No	No	No	No	No	No	No	No
	.	-		7		!	l																

Table B9. Returns to Skills: Robustness Analysis. Correcting for Selection in the Labor Market

					0							
	Endowments	Coefficients	Endowments	Coefficients	Endowments	Coefficients	Endowments	Coefficients	Endowments	Coefficients	Endowments	Coefficients
Totals	0.044	-0.174	0.0628	-0.193	0.054	-0.184	0.0589	-0.19	0.0703	-0.2	0.0176	0.148
	[0.0166]***	[0.0230]***	[0.0156]***	[0.0237]***	[0.0160]***	[0.0232]***	[0.0156]***	[0.0237]***	[0.0156]***	[0.0239]***	[0.0169]	[0.0239]***
Other factors	0.0548	-0.696	0.0615	-0.734	0.0569	-0.701	0.0629	-0.673	0.063	-0.709	0.0586	-0.69
	[0.0147]***	[0.241]***	[0.0152]***	[0.245]***	[0.0147]***	[0.242]***	[0.0152]***	[0.246]***	[0.0153]***	[0.245]***	[0.0150]***	[0.242]***
ICT skills	-0.0108	-0.0118										
Readiness to learn	[1 2000.0]	[1000.0]	0.00129	-0.0345								
			[0.00256]	[0.127]								
Management and communication					-0.00292 [0.00500]	0.00127 [0.0437]						
Self-organization							-0.00394	0.0337				
Marketing and accounting							[0.00295]	[0.0040]	0.00731	-0.0841		
)									[0.00284]***	[0.0363]**		
STEM-quantitative										1	-0.041 0.006671***	-0.104 In n3941***
Constant		0.534		0.576		0.516		0.449		0.593	[00000.0]	0.647
		[0.243]**		[0.265]**		[0.244]**		[0.252]*		[0.252]**		[0.246]***
Observations	3,242	3,242	3,242	3,242	3,242	3,242	3,225	3,225	3,242	3,242	3,242	3,242
Countries FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Source: Own elaboration be	ased on PIA	AC and ST	EP surveys	·								
Notes: Robust standard erro	ors in brack	ets. *** p<(0.01, ** p<	0.05, * p<().1							

Table B10. Oaxaca-Blinder Decomposition of the Gender Wage Gap

Appendix C





Source: Own elaboration based on PIAAC and STEP surveys.































Coefficients

Endowments

55

Coefficients

Endowments